### HOW DO CHANGES IN FINANCIAL REPORTING STANDARDS AFFECT RELATIONSHIP LENDING?

# 2024

BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2437

Daniel Dejuan-Bitria, Wayne R. Landsman, Sergio Mayordomo and Irene Roibás HOW DO CHANGES IN FINANCIAL REPORTING STANDARDS AFFECT RELATIONSHIP LENDING?

#### HOW DO CHANGES IN FINANCIAL REPORTING STANDARDS AFFECT RELATIONSHIP LENDING? <sup>(1)</sup>

#### Daniel Dejuan-Bitria

BANCO DE ESPAÑA

#### Wayne R. Landsman

KENAN-FLAGLER BUSINESS SCHOOL, UNIVERSITY OF NORTH CAROLINA AT CHAPEL HILL

#### Sergio Mayordomo

BANCO DE ESPAÑA

#### Irene Roibás

BANCO DE ESPAÑA

Documentos de Trabajo. N.º 2437 October 2024

NumeroDOI

<sup>(\*)</sup> E-mail addresses: daniel\_dejuan@bde.es, wayne\_landsman@kenan-flagler.unc.edu, sergio.mayordomo@bde. es, irene.roibas@bde.es. We thank Jorge Abad, Roberto Blanco, Robert Bushman, Maria Dimou, Javier Górmez-Biscarri, Yunzhi Hu, Germán López-Espinosa, José Luis Peydró, Amedeo Pugliese, María Rodríguez-Moreno and seminar participants at the Banco de España, the University of North Carolina at Chapel Hill and the 8th Research Workshop Task Force on Banking Analysis for Monetary Policy (Bank of Greece, Athens) for their useful comments. Daniel Dejuan-Bitria gratefully acknowledges financial support from the US-Spain Fulbright grant and AGAUR (FI 2022). Wayne R. Landsman thanks the Kenan-Flagler Business School for financial support. The views expressed in this paper are those of the authors and do not necessarily represent the views of the Banco de España or the Eurosystem.

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2024

ISSN: 1579-8666 (on line)

#### Abstract

This paper analyses the effect of the expected credit loss model under IFRS 9 on relationship lending in Spain. We document that relationship exclusivity between a bank and a firm has a positive effect on the growth of credit. However, this positive effect is significantly reduced after implementation of IFRS 9. We estimate that in 2018 the negative impact of IFRS 9 on relationship lending led to a reduction in credit to Spanish non-financial firms of 2.8% of their total outstanding credit, suggesting a sizeable effect on the availability of credit. For borrowers with Stage 1 loans, we show that the new regulation has a negative impact on relationship lending at firms with a higher probability of default and whose credit quality has deteriorated. Our findings are consistent with a change in the incentives that underpin relationship lending.

Keywords: relationship lending, IFRS 9, credit, probability of default.

JEL classification: D82, G21, G28.

#### Resumen

En este documento se examina el efecto sobre el crédito relacional en España de la implementación, en enero de 2018, del modelo de pérdida crediticia esperada según las Normas Internacionales de Información Financiera (NIIF-9). Los resultados obtenidos muestran que la relación de exclusividad entre un banco y una empresa, definida como aquella existente cuando la mayor parte del crédito de una empresa es concedido por un determinado banco, tiene un efecto positivo en el crecimiento del crédito. Sin embargo, este efecto positivo se reduce significativamente tras la implementación de NIIF-9. Se estima que el impacto negativo asociado a esta norma contable sobre el crédito relacional ha conducido a una reducción del saldo vivo de los préstamos concedidos a las empresas no financieras españolas del 2,8 %. Un análisis adicional, que restringe la muestra a empresas cuya totalidad del crédito se podría clasificar en la etapa 1 de acuerdo con la norma contable, ilustra que el impacto negativo de la nueva regulación sobre los préstamos relacionales se observa para las empresas con una mayor probabilidad de impago de su deuda bancaria y cuya calidad crediticia se ha deteriorado de manera significativa. Estos resultados son coherentes con un cambio en los incentivos que sustentan el crédito basado en las relaciones banco-cliente.

Palabras clave: crédito relacional, NIIF-9, crédito, probabilidad de impago.

Códigos JEL: D82, G21, G28.

#### 1. Introduction

The questions this study addresses are whether and how bank regulation affects the role of relationship banking in facilitating credit access for firms. Although there has been ample research on the impact of banking regulations on the supply of credit (e.g., Jiménez et al., 2017; Gropp et al., 2019), and on the determinants of relationship lending (e.g., Sette and Gobbi, 2015; Bolton et al., 2016; Banerjee et al., 2021), there is limited knowledge on whether regulatory changes affect relationship banking and the channels through which this effect can occur. These are relevant public policy questions given the significant role of relationship lending in facilitating borrowers' access to capital through loan contracting over time (e.g., Degryse et al., 2015).

To address our research questions, we exploit an accounting standard change in 2018, the implementation of IFRS 9 by Spanish banks. This is a powerful setting to assess the impact of bank regulation on relationship lending for two reasons. First, the new accounting standard is highly material for the banking industry (e.g., GPPC, 2017).<sup>1</sup> IFRS 9 mandates a forward-looking approach to estimate loan loss provisioning, the most important accrual for banks, based on expected credit losses (ECL). This is a major departure from the previous methodology that relied on incurred credit losses (ICL) and is likely to result in a significant increase in recognized credit losses, which can have a significant effect on bank regulatory capital. Second, unlike other banking regulations introduced gradually and spanning several years (e.g., Basel III), the shift to IFRS 9 was entirely effective with the fiscal year change in January 2018. By comparing the role of relationship lending using a short window before and after the implementation of IFRS 9, it is less likely that our results are affected by other aggregate factors.

A strong relationship between a bank and a borrower is characterized by both the access to soft information about the borrower and a higher level of exposure between the two entities (e.g., Kysucky and Norden, 2016; López-Espinosa et al., 2017). Prior literature reveals that these two characteristics of relationship lending generally increase the availability of credit to borrowers (e.g., Boot, 2000; Kysucky and Norden, 2016). First, the use of soft information by banks may be key to mitigate information frictions between the lender and the borrower (e.g., Petersen and Rajan, 1994). Second, the "exclusivity dimension" of relationship banking implies that a bank that has an exclusive relationship with a borrower (i.e., a large share of the borrower's total credit) has a greater incentive to facilitate new lending to avoid the recognition

<sup>&</sup>lt;sup>1</sup> The Global Public Policy Committee states: "The introduction of the requirement to estimate expected credit losses ("ECL") under IFRS 9 Financial Instruments is perhaps the single most significant change in the history of financial reporting of banks" (GPPC, 2017).

of credit losses, because the decision to reduce lending could increase the probability of the borrower defaulting (e.g., Boot, 2000).<sup>2</sup>

We predict and find that IFRS 9 has a negative impact on relationship lending by affecting the "exclusivity dimension" of relationship banking, thereby reducing the availability of credit to borrowers. Prior to IFRS 9, under the ICL model, banks could finance riskier borrowers under an exclusive relationship without any direct impact on loan loss provisions. Loan loss provisions, which decrease regulatory capital, were only recognized if there was objective evidence of impairment or a triggering event. In this case, banks were willing to facilitate new credit to riskier relationship borrowers to prevent default and the recognition of credit losses (Boot, 2000). Under IFRS 9, loan loss provisions are forward-looking and recognized at loan origination. Thus, banks may anticipate having to recognize loan loss provisions when extending new credit and during the life of the loan for borrowers with riskier profiles, even if there is no evidence of impairment. After implementation of IFRS 9, an exclusive relationship with a risky borrower likely reduces the bank's incentive to increase that share. For this reason, we predict that IFRS 9 decreased the positive effect of holding an exclusive relationship on the availability of credit.

To examine the impact of IFRS 9 on relationship lending, we analyse the lending decisions of Spanish banks over the period June 2016 to June 2019, a window that spans three semesters before and after the introduction of the ECL model in January 2018. Specifically, we analyse growth in credit provided to borrowers in banking relationships. Our focus on Spanish banks provides a powerful setting to study the impact of IFRS 9 on relationship lending because of the relatively large fraction of Small and Mediums-sized Enterprise (SME) borrowers in Spain.<sup>3</sup> SME lending is characterized by high information frictions between borrowers and lenders and difficulty for borrowers to access external finance. Unlike in other European countries, where only listed banks are required to adopt IFRS 9, Spanish regulation mandates all—listed and non-listed—banks to implement IFRS 9, regardless of whether they have issued securities in a regulated market (Circular 4/2017).<sup>4</sup> Also, the Spanish economy was experiencing significant financial stability when the new accounting standard was first introduced. This alleviates the potential confounding effect of a crisis or a general macroeconomic downturn on relationship lending (Sette and Gobbi, 2015; Bolton et

<sup>&</sup>lt;sup>2</sup> Relationship lenders can have incentives to continue to provide credit to poorly performing borrowers even in the absence of bank regulation. For example, in a theoretical model, Hu and Varas (2021) shows that relationship lenders will continue to provide credit to so-called "zombie" firms—firms whose operating cash flows are consistently less than their interest payments—if the firms will soon payoff their loans by substituting market financing for private financing with relationship lenders.

<sup>&</sup>lt;sup>3</sup> Small and Medium-sized Enterprises are defined by the European Commission as companies with fewer than 250 employees and an annual turnover of no more than €50 million, or an annual balance sheet total of no more than €43 million. <sup>4</sup> Circular 4/2017 issued by the Banco de España, required the adoption of the expected credit loss model from IFRS 9 in 2018 for all credit institutions in Spain, including both listed and non-listed banks. The adoption of IFRS 9 became mandatory for all listed banks in the EU in that same year.

al., 2016; Banerjee et al., 2021). Finally, our focus on Spain allows us to access the rich loan-level data from the Credit Register of the Banco de España.

To assess the effect of IFRS 9 on relationship lending, we employ an OLS regression to compare the growth of credit for a borrower from a bank with a previous strong relationship relative to a bank with a weaker relationship, before and after the implementation of IFRS 9. We perform our analyses using two different proxies for relationship lending that reflect the exclusivity of a relationship between a bank and a firm commonly employed in prior research (e.g., Elsas, 2005; Kysucky and Norden, 2016; López-Espinosa et al., 2017). We consider the perspective of the borrower, i.e., whether a firm has a high proportion of its credit in a given bank, and the perspective of the lender, i.e., whether a bank has exposure to a given firm. Additionally, as a robustness check, we use the number of new relationships between a lender and a borrower over the past five years as a proxy for relationship lending. We include firm-time fixed effects to account for credit demand trends (Khwaja and Mian, 2008), and bank-time fixed effects as controls for concurrent bank-specific time varying factors.

IFRS 9 requires banks to recognize expected credit losses for financial instruments based on a three-stage classification system. New originated loans and loans without a significant increase in credit risk since origination are classified in Stage 1. For these loans, banks recognize expected credit losses considering the probability that the borrower defaults in the following 12 months. Loans that have experienced a significant increase in credit risk since initial recognition but have not yet incurred a credit loss event are classified in Stage 2. In this case, expected credit losses are based on the probability of default (PD) over the entire life of the instrument. Finally, non-performing loans, with a PD close to 100%, are classified in Stage 3. Provisioning under the ICL methodology was restricted and akin to the estimation of the accrual for loans classified in Stage 3 under IFRS 9. We expect the effects of IFRS 9 on relationship lending to be strongest for loans with higher credit risk in Stages 1 and 2, and weakest for borrowers whose loans are in Stage 3 because the new regulation does not change the loan loss allowance associated with loans that have objective evidence of impairment.

A key challenge to estimate the effect of IFRS 9 on relationship lending is that the three-stage asset classification under IFRS 9 did not exist before the introduction of the new regulation and data were not available immediately following its introduction for Spanish banks.<sup>5</sup> We address this limitation by proposing an approximation of Stage 1 at the borrower level based on observable borrower characteristics before and after the introduction of IFRS 9. A borrower is classified in Stage 1 if it

<sup>&</sup>lt;sup>5</sup> Furthermore, IFRS 9 requires a determination of the expected credit losses and the stage classification at the loan level. As described below, we approximate the stages at the borrower level because unavailability of loan-level credit rating information for our sample borrowers, which predominantly consists of micro, small, and medium-sized enterprises.

satisfies two criteria. First, we require that all its loan exposures are performing. Second, we require that the borrower has not experienced a significant increase in its credit risk, potentially triggering a transition of the outstanding credit from Stage 1 to Stage 2.<sup>6</sup> We estimate and use the change in the PD of a firm in the years 2014 and 2015 to measure the variation of credit risk of a borrower in the pre-IFRS 9 period. Similarly, we use the change in a firm's PD during 2016 and 2017 to measure the increase of credit risk in the post-IFRS 9 period. We conduct our main analysis focusing on firms in Stage 1 because this is the stage that we can approximate with higher accuracy (see Section 2). In supplementary analysis, we evaluate whether the effects on relationship lending are more pronounced for firms whose credit risk has deteriorated sufficiently to transition to Stage 2 and whether there is any effect for firms with all outstanding credit classified as non-performing, i.e., those in Stage 3.

Our baseline analysis indicates that, on average, IFRS 9 reduced the role of relationship lending in facilitating firms' access to credit. Nonetheless, there is still a positive net effect of relationship lending on loan growth after IFRS 9, which indicates that it remains an important mechanism for access to credit. Considering relationship lending from the borrowers' (banks') perspective, we show that the credit growth of relationship borrowers was 13.2 (8.5) pp higher before IFRS 9, relative to non-relationship lending borrowers. This difference decreased by 2.1 (2.5) pp in the aftermath.<sup>7</sup> Thus, relationship banking plays a smaller role in facilitating new lending under the new accounting standard. In addition, we show that this effect implies a reduction in the availability of credit at the firm level. Firms that relied more on relationship banking were unable to substitute the lending from banks with a previous strong relationship to banks with a weaker relationship. An approximate calculation suggests a sizeable effect on the availability of credit. In particular, the negative effect of IFRS 9 on relationship lending implies a reduction in credit of €13.4 billion in 2018, which represent 2.8% of the total outstanding credit of Spanish non-financial corporations as of December 2017.

When a loan is secured by collateral, transitioning from the ICL model to the ECL model may not have a significant effect on loan loss provisions because collateral reduces the loss given default and expected credit losses. Hence, banks lending to relationship borrowers whose credit is guaranteed by collateral do not expect to raise high loan loss provisions at loan origination or throughout the loan's life and this applies both before and after IFRS 9 implementation. Given that the new impairment model has minimal impact on loan loss provisions for borrowers with

<sup>&</sup>lt;sup>6</sup> IFRS 9 does not provide a sharp rule regarding what should be considered a significant increase in credit risk, triggering a transition from Stage 1 to Stage 2. For this reason, we employ the existing guidance of regulators and consider a threefold increase in the credit risk of a borrower to be a sufficient condition for a significant increase in credit risk (see Section 2.2). To the extent that our approach leads to classification error, it will be harder for us to detect the effect on relationship lending of a firm transitioning from Stage 1 to Stage 2.

<sup>&</sup>lt;sup>7</sup> As shown below, the inferences we draw are robust to alternative definitions of the credit growth, relationship lending, sample selection criteria and the possibility of an anticipation effect by banks to IFRS 9.

guaranteed credit exposures, we find that the introduction of IFRS 9 does not alter the effect of relationship banking on credit growth.

Next, we investigate heterogenous effects of IFRS 9 on relationship lending based on borrower characteristics. First, we predict that the effect of IFRS 9 on relationship lending is greater for borrowers with a riskier profile. The cost in terms of regulatory capital is larger for loans to riskier borrowers because of the higher level of provisions at loan origination. In addition, a higher *exante* risk may increase the likelihood of loan migration from Stage 1 to Stage 2 in the future, which considers lifetime default probabilities, resulting in a significant rise in loan loss provisions.<sup>8</sup> To this end, we exploit the cross-sectional variation in the risk of borrowers whose loans are classified in Stage 1. In particular, we repeat our baseline regression partitioning the sample into low and high risk borrowers. Our findings show prior to IFRS 9, holding a relationship had a similar effect on the growth of credit for low and high risk borrowers. However, following implementation of the new standard, we find that the ECL model has a negative impact on relationship banking for high risk borrowers but not for low risk ones. These findings suggest that the introduction of IFRS 9 created a new asymmetry regarding the role of relationship lending, whereby riskier borrowers benefit less from relationship lending facilitating access to credit.

Second, we consider the role of the deterioration of the credit quality of a borrower. New credit to borrowers that have already experienced a significant increase in credit risk are more likely to migrate from Stage 1 to Stage 2 when lifetime PDs are considered, and provisions may significantly rise (ESRB, 2017). To this end, we partition the sample of borrowers in Stage 1 into low risk firms whose credit risk has remained low and high risk firms with a deterioration of credit quality, which we approximate as the variation of the PD in the previous years. We show that the negative impact of IFRS-9 is present for high risk borrowers in Stage 1 whose credit quality has deteriorated but not for low risk firms in Stage 1 with stable credit quality.

Third, we further investigate how IFRS 9 affects relationship lending for Stage 2 borrowers by focusing on risky firms that have experienced a three-fold increase in credit risk—the measure that is used by Spanish bank regulators as a significant decrease in credit quality. These borrowers are not included in the sample of our main analysis relating to firms that we approximate as being Stage 1 borrowers and are likely to have outstanding credit in Stage 2 (see Section 2.2). We find that the negative effect of IFRS 9 on relationship lending is significantly larger for these firms, indicating that banks have greater incentives to reduce their exposure to relationship borrowers if there is a high probability of the loan downgrading to Stage 2 in future periods.

<sup>&</sup>lt;sup>8</sup> Throughout the paper we use the term "ex-ante risk" of a borrower to refer to the credit risk level of a firm before obtaining new credit.

Finally, we conduct a falsification test to mitigate the likelihood that an omitted variable may affect the role of relationship lending for riskier borrowers after 2018, thereby confounding our inferences regarding the effect of IFRS 9. If this were the case, we would expect the omitted variable to affect the role of relationship lending for Stage 3 borrowers, which are highly risky firms, i.e., firms with all their credit exposures being impaired. In contrast, IFRS 9 should not affect the role of relationship lending for this profile of borrowers because the ICL model already contemplated raising life-time provisions at loan origination when there was objective evidence of impairment. Therefore, we analyse the effect of the ECL model on relationship lending for this group of firms. In particular, we document a large effect of relationship lending on the growth of credit both before and after IFRS 9. This result increases our confidence that our tests correctly identify the effect of IFRS 9 on relationship lending.

Our analysis relates to the literature focusing on the value and consequences of relationship lending (e.g., Petersen and Rajan, 1994; Berger and Udell, 1995; Degryse et al., 2015; Banerjee et al., 2021). To the best of our knowledge, our study is the first to consider the effect of banking regulation on relationship lending.<sup>9</sup> We focus on the implications of the new impairment model under IFRS 9 introduced in the year 2018. Our findings suggest that holding an exclusive relationship facilitates firms' access to external finance, both before and after the regulatory change. However, the increase in the cost of lending after IFRS 9 reduces the incentives of banks to grant new credit to relationship borrowers that are *ex-ante* riskier and whose credit quality has deteriorated.

Our paper also contributes to the literature that studies the effect of loan loss provisioning methodologies on bank's lending behaviour and risk taking (e.g., Beatty and Liao, 2011; Bushman and Williams, 2012; 2015; Abad and Suarez, 2018; Morais et al., 2020). Focusing on the introduction of IFRS 9, Bischof et al. (2022) finds that German banks strategically adjust their internal ratings and lending decisions to minimize loan loss provisions. The study shows that the impact of IFRS 9 on the supply of credit depends on the risk profile of borrowers.<sup>10</sup> We contribute to this literature in three ways. First, we focus on the interaction between the new impairment model and relationship lending. This focus is particularly relevant because relationship lending plays a key role in facilitating access to credit for significant market segments, such as private firms and SME borrowers. SMEs represented 99.7% of non-financial Spanish companies as of January

<sup>&</sup>lt;sup>9</sup> There is a large literature that studies the impact of bank regulation on lending (see, e.g., Jiménez et al. (2017) and Gropp et al. (2019) for recent empirical contributions). However, the effect of prudential or accounting regulation on relationship lending has not yet been explored.

<sup>&</sup>lt;sup>10</sup> Overall, prior research suggests that the implementation of IFRS 9 affects banks' lending and reporting decisions (e.g., Ertan, 2021; López-Espinosa et al., 2021).

1, 2023, and are attributed with 40.7% of the total gross fixed capital formation (see Banco de España, 2024). SMEs are also characterized by high information frictions and bank credit is the main source of external financing for this type of firms. This is why relationship lending is crucial for their access to credit and for their investment activity. Second, we develop a methodology for approximating the three stages under IFRS 9 at the borrower level that can be used in future research. Our analysis includes micro, small, and medium-sized firms because our classification of stages is not limited to firms with a loan volume above a certain threshold. In contrast, Bischof et al. (2022) focuses on large firms because the study's sample is restricted to loans with a volume of more than two million euros in at least one quarter of the sample period. This allows us to study the role of relationship lending for micro, small, and medium-sized firms, where it is expected to be more relevant. Finally, we examine the impact of collateral under IFRS 9, where the estimation of loan loss provisions depends on the expected loss given default, which can be significantly reduced by the presence of collateral.

The remainder of the paper is organized as follows. Section 2 describes the institutional background for our analysis and our approach for approximating our classification at the borrower level the three stages under IFRS 9. Section 3 describes our dataset and the variables employed in the analysis. Section 4 presents our research design. Section 5 discusses the main results, focusing on borrowers classified in Stage 1 under IFRS 9. Section 6 explores the effect of the transition to Stage 2. Section 7 presents findings from the falsification test. Section 8 presents finding from additional robustness tests of our baseline regression and Section 9 concludes.

#### 2. Institutional Background and Research Design

#### 2.1 The new impairment model under IFRS 9

In contrast to the incurred loss (ICL) model that was prevalent before 2018, the new provisioning methodology under IFRS 9, the expected credit loss (ECL) model, requires banks to recognize loan loss provisions at loan origination or purchase, reflecting expected credit losses. The ECL model, which is forward-looking in nature, was introduced in response to the "too little, too late" concern associated with ICL model, particularly during financial downturns such as the 2008 financial crisis (e.g., Gaston and Song, 2014; Acharya and Ryan, 2016; de Haan and van Oordt, 2018). The ICL model mandates banks to recognize provisions under verifiable evidence of impairment. As a result, loan losses are primarily considered when the probability of default (PD) is close to 100% (Novotny-Farkas, 2016). The ECL model eliminates the "loss event" constraint of the old methodology by considering loan losses when PDs are less than 100%.

A key input for the estimation of expected credit losses is the probability that a borrower defaults in the future. Under IFRS 9, banks must estimate the PD considering different

macroeconomic scenarios. Furthermore, the new accounting standard mandates banks to classify financial assets in three stages, according to the credit quality deterioration since initial recognition. Such classification determines the time horizon for the estimation of the PD. New loans and loans without a significant increase in credit risk (SICR) are allocated in Stage 1, where only PDs in the next 12 months are considered. If a loan suffers a SICR, a migration to Stage 2 occurs and the allowance must be updated to incorporate life-time PDs. Finally, loans that are non-performing are classified in Stage 3. Hence, Stage 3 loans result in essentially the same loan loss allowance as those recognized under the ICL model.

Figure 1 illustrates the loan loss allowance (y-axis) associated with a financial instrument under both the ICL methodology and the ECL model of IFRS 9 as the credit quality of the instrument deteriorates (x-axis). The diagram depicts sharp differences between the two accounting regimes. First, the pronounced convexity of provisions under the ICL model (red line) reflects the concern of prudential regulators. No matter the *ex-ante* credit risk of a borrower or the deterioration of credit quality, provisions are not recognized until there was objective evidence of impairment as of the balance-sheet date.

In contrast, the ECL model under IFRS 9 yields a smoother pattern of loan loss provisions as credit quality deteriorates (blue line). At origination, before the loan has suffered any change in its credit risk, banks must already estimate the PD in the next 12 months of the borrower and expected credit losses in the next 12 months, and recognize a provision for loan losses. This is reflected in the intersection of the allowance scheme with the y-axis in the origin (point A in the Figure 1). As the credit quality of the loan deteriorates, banks must update the level of the accrual. Notably, a sharp increase occurs when a loan migrates from Stage 1 to Stage 2 as a result of consideration of lifetime PDs in this second bucket, i.e., after a SICR occurs, as depicted in Figure 1. The literature has referred to the transition between the first and the second buckets as a "cliff" caused by the sharp increase in provisions (ESRB, 2017). Finally, a loan that becomes non-performing is allocated in Stage 3 and deemed credit impaired, with PD=100%, thereby coinciding with the ICL methodology.

#### 2.2 Identifying stages at the borrower level

Ideally, to estimate the impact of the ECL model under IFRS 9 on relationship banking we would control for the classification of loans in the different stages and aggregate this information at the borrower level. Each stage considers different horizons for the estimation of PDs. This may affect substantially the estimated expected credit losses. Controlling for the stage classification in our empirical framework would require two conditions. First, it would be necessary to identify correctly each loan's as-if stage before the implementation of IFRS 9. Second, it would be

necessary to observe the actual stage allocation of each loan after the introduction of the rule. The distinction of the buckets would allow for the correct classification of the outstanding credit of a borrower. In this case, we would be able to identify the distinct role of the stages in contributing to the effect of the new rule.

The lack of stage classification prior to IFRS 9 and the fact that, immediately afterwards, banks do not disclose the classification at the loan or borrower level poses a challenge to our identification strategy. We address this challenge by proposing an approximation of the stages at the borrower level using observable information before and after the introduction of the new rule. Our main analysis focuses on firms whose loans are classified in Stage 1, in light of the fact that identifying when a loan is in Stage 2 cannot be made with precision. Moreover, the ample room for discretion and professional judgement in the application of IFRS 9 makes such precise imputation even more difficult.<sup>11</sup> However, it is still possible to estimate the likelihood that the newly issued credit initially classified in the first stage may migrate to the second bucket by considering the determinants of this transition.

#### 2.2.1 Identifying Stage 1

#### Stage 1. Baseline sample.

We classify a borrower in Stage 1 if it satisfies two criteria. First, all its loan exposures are performing. Second, it must not be a high yield firm that has experienced an annual three-fold increase in credit risk, considering the change in credit risk in 2015 for the pre-IFRS 9 period and in 2017 for the post-IFRS 9 period. This latter restriction is designed to eliminate borrowers whose outstanding credit may have already migrated from Stage 1 to Stage 2, and is consistent with the criterion used by regulators as an indicator of a significant increase in credit risk (see, e.g., the AQR Manual and the EBA Stress Test Methodological guidance).<sup>12</sup>

Our approach for identifying Stage 1 borrowers requires the estimation of the credit risk of a firm such that we can classify it as an investment grade or high yield company. To do this, we

<sup>&</sup>lt;sup>11</sup> Supervisors provide general guidance to banks regarding what should be deemed a significant credit quality deterioration (e.g., a threefold increase in the lifetime PD from initial recognition). However, the standard stages that the SICR triggers should not be rule-based but rather rely on professional judgment and on all relevant quantitative and qualitative information. A more precise identification of Stage 2 would be possible if we could observe the internal credit rating that banks assign to their borrowers, as suggested by Bischof et al. (2022). From each bank's Basel III Pillar 3 disclosures, the study derives the individual mapping from PDs into internal ratings. However, we cannot apply this approach because the Spanish credit registry only contains information about the internal risk assessments of banks (i.e., borrowers PD) from 2019 onwards, and they are only available for lenders using an internal rating-based (IRB) approach.

<sup>&</sup>lt;sup>12</sup> The AQR (Asset Quality Review) Manual is a guide used by banking regulators to assess a bank's asset quality and identify risks in its balance sheet. It provides a standardized methodology for evaluating credit, market, and operational risks. The EBA Stress Test Methodological Guidance is a common framework for stress testing EU banks. It outlines the methodology, scenarios, and templates that banks must use to assess their resilience to different stress scenarios.

estimate annually the PD of each firm following the methodology of Blanco et al. (2023) (see section 3.2). We use the PD of a firm in the years 2014 and 2015 to estimate the level and change of risk of a borrower in the pre-IFRS 9 period. Similarly, we use the PD of a borrower in the years 2016 and 2017 to measure the variation and level of risk of a firm in the post-IFRS 9 period. We use the PD intervals of the European Central Bank's CQS methodology to define high yield firms (PD>0.4%) and investment grade firms (PD $\leq$ 0.4%).<sup>13</sup>

#### Partition of Stage 1 at the borrower level: Ex-ante risk and credit quality deterioration

The effect of IFRS 9 on relationship lending may be heterogenous for different borrowers within Stage 1. We predict that the negative effect of IFRS 9 on relationship lending is greater for firms with a riskier profile. Before IFRS 9, a bank with an exclusive relationship with a risky borrower (i.e., holding a large share of the borrower's total credit) had strong incentives to facilitate new lending to avoid recognizing credit losses, because reducing lending to this borrower could increase default risk (Boot, 2000). Under IFRS 9, loan loss provisions are based on expected credit losses and are recognized at loan origination. This may reduce the bank's incentive to increase lending to a risky borrower in an exclusive relationship. In contrast, relationship lending to less risky firms may not be affected by IFRS 9 because expected credit losses for these borrowers may be low.

To examine the heterogeneous effects of IFRS 9 on relationship lending for both relatively low and high risk borrowers, we partition our Stage 1 sample as follows. First, we partition the sample into high yield and investment grade firms (i.e., borrowers with PD above or below 0.4%, respectively). We expect the effect of IFRS 9 on relationship lending to be stronger for high yield firms. Second, we further refine this partition by categorizing Stage 1 firms into high yield firms whose credit quality has deteriorated (i.e., the PD has increased in the previous years), and investment grade firms with a stable credit quality (i.e., the PD has remained equal or below 0.4%).<sup>14</sup> This refinement allows us to identify more precisely two groups of firms that may be differentially affected by IFRS 9.<sup>15</sup> We expect the impact of IFRS 9 on relationship lending to be stronger for high-yield firms whose credit quality has deteriorated.

<sup>&</sup>lt;sup>13</sup> The Credit Quality Steps (CQS) methodology is used by the European Central Bank to assess the creditworthiness of companies in the Europystem credit operations.

<sup>&</sup>lt;sup>14</sup> Specifically, the first group is comprised of firms with a PD above 0.4% in 2015 (for the pre-IFRS 9 period) and in 2017 (for the post-IFRS 9 period) and this PD has increased with respect to the previous year (i.e., 2014 and 2016, respectively). The second group is comprised of firms with a PD below or equal to 0.4% in 2014 and 2015 (for the pre-IFRS 9 period) and in 2016 and 2017 (for the post-IFRS 9 period).

<sup>&</sup>lt;sup>15</sup> The outstanding credit of borrowers that have experienced an increase in credit risk is more likely to migrate from Stage 1 to Stage 2 where lifetime PDs are considered, leading to a significant rise in provisions (ESRB, 2017). We expect the effect of IFRS 9 on relationship lending to be stronger for this group of firms. Conversely, we expect the effect to be non-significant for safe firms with stable credit quality.

#### 2.2.2 The transition to Stage 2

The effect of IFRS 9 on relationship lending may be more pronounced for borrowers that have experienced a significant increase in credit risk (i.e., borrowers with outstanding credit classified in Stage 2). The probability that new credit that initially is in Stage 1 downgrades to Stage 2 may be higher for this profile of firms. Under IFRS 9, new lending to these borrowers may imply provisioning for large expected credit losses not only at origination, but also if the new loan downgrades to Stage 2 in the future, where lifetime PDs are considered.

To examine the role of Stage 2, we define a new sample of firms. We define Stage 2 borrowers as high-yield firms that have experienced a threefold annual increase in credit risk. Specifically, we consider firms with a PD above 0.4 percent in 2015 (for the pre-IFRS 9 period) and 2017 (for the post-IFRS 9 period), and for which the PD has experienced a threefold increase compared to the previous year (i.e., in 2014 and 2016, respectively). These borrowers were excluded from the sample of our main analysis (our approximation of Stage 1 borrowers).

#### 2.2.3 Identifying Stage 3

Finally, we consider borrowers whose outstanding credit may be allocated in Stage 3. The accounting treatment of impaired firms in the ICL and ECL methodology requires banks to consider a PD of 100% to estimate expected losses. We consider borrowers to be classified in Stage 3 if all its credit exposures have a non-performing status. The definition of Stage 3 allows us to conduct a falsification test, because the new impairment model does not introduce major differences for this profile of firms. Thus, we predict no effect of IFRS 9 on relationship lending and hence access to credit for borrowers in Stage 3.

#### 3. Data and Variables

#### 3.1 Data

To study the impact of IFRS 9 on relationship lending, we combine three different data sources: the Spanish Credit Register (CIR), firm balance-sheet and profit and loss accounts from the Banco de España's Central Balance Sheet Data Office, and supervisory bank balance-sheet information.

The CIR contains monthly information on all outstanding loans to non-financial firms granted by all credit institutions operating in Spain. Our dataset provides a comprehensive representation of the population of loans to firms. From the CIR we obtain detailed information about each loan, including the amount (both drawn and undrawn), which will be key to construct our main dependent variable, and its performing situation. Focusing on the period from June 2016 to June 2019, we aggregate the outstanding amount of credit of each firm in each bank at the end of each semester to obtain the total bank-firm credit exposure, which also allows us to construct our different proxies of relationship banking. Additionally, the database provides information about the borrower and bank identity, which permits us to match each loan with characteristics of the bank and the firm. We take into account all mergers and acquisitions of banks that took place between 2016 and 2019. Specifically, if a bank was acquired during a given semester, we consider that this bank was the same entity as the acquiring bank in the previous semester by creating a synthetic bank. By doing so, we ensure that the variation of credit is not affected by the disappearance of banks or the creation of new banks arising from mergers and acquisitions.

We complement the information from the Spanish Credit Register with annual firm-level data from the Banco de España's Central Balance Sheet Data Office (CBSDO) to estimate the credit risk of a firm (i.e., the probability of default, PD). This database provides information on the financial accounts of more than 750,000 non-financial corporations on an annual basis by combining administrative data at the annual level from accounts filed with the mercantile registries in Spain with information compiled from surveys of non-financial firms conducted by the Central Balance Sheet Office of Banco de España<sup>16</sup>. We match this firm-level information with loan-level data using the unique firm fiscal identifier. Lastly, we obtain bank financial statement information collected by the Banco de España in its role of banking supervisor. We define annual firm and bank characteristics as of December 2015 when referring to the pre-IFRS 9. For the post-IFRS 9 period, we use firm and bank information as of December 2017.

Our analyses are restricted to firm-bank pairs with an existing relationship at the beginning of each semester for which we can calculate the growth of credit and for borrowers with enough information to estimate their PD.<sup>17</sup> This approach for defining an existing relationship, which closely follows Sette and Gobbi (2015), eliminates the bias that would result from defining the growth of credit for firms with zero initial outstanding credit. Namely, a mechanically positive association would arise between firms with zero initial outstanding credit (and, thus, a very low relationship strength with the considered bank) and the growth of credit. Finally, to focus on borrowers in Stage 1 in our baseline analysis, we restrict our sample to firms with all credit exposures classified as performing, excluding those high yield firms that have experienced a threefold increase in their PDs (see subsection 2.2.1). The sample that we use in our main analysis, reflecting borrowers with loans in Stage 1, consists of 2,269,993 firm-semester observations from

<sup>&</sup>lt;sup>16</sup> We restrict our sample to firms with adequate accounting quality based on the Banco de España's internal classification criteria. In particular, we apply two filters provided by the CBSDO and exclude firms with (i) balance sheets with non-reliable monetary units; and (ii) firms with inadequate information in their financial statements (with blatant accounting errors, such as large mismatches in balance sheet amounts, negative values in items that should be positive by definition, missing headings, or figures of disproportionate magnitude).

<sup>&</sup>lt;sup>17</sup> We follow the methodology of Blanco et al. (2023) to estimate the PD of a borrower. To measure the level and change in risk of a borrower before (after) the IFRS 9 period, we estimate the probability of default (PD) of a firm for the years 2014 and 2015 (2016 and 2017). See Section 3.2 for more details.

the second semester of 2016 to the first semester of 2019 that are distributed in a range between 618,489 and 711,962 observations per semester.

#### 3.2 Variables

The dependent variable in our analyses is the semi-annual credit growth, measured at the bankfirm level. Namely, we consider the change in credit balance of firm f in bank b in a given semester with respect to the previous semester divided by the average credit balance in the two semesters.<sup>18</sup>

To measure relationship lending, we focus on the exclusivity of a relationship between a bank and a firm. This variable has been shown to be a key determinant of relationship banking and plays a crucial role in facilitating access to credit. A bank with an exclusive relationship with a borrower (i.e., a large share of the borrower's total credit) has a greater incentive to facilitate new lending to avoid the recognition of credit losses, because the decision to reduce lending could increase the probability of the borrower defaulting (e.g., Boot, 2000). Supporting the significance of exclusivity, López-Espinosa et al. (2017) compares the importance of relationship exclusivity with other relationship variables and find that exclusivity has the greatest economic impact in facilitating access to credit in Spain. Similarly, Kysucky and Norden (2016) shows that a higher exclusivity is associated with lower loan rates. Furthermore, Elsas (2005), which identifies relationship lending based on the self-assessment of banks, finds that a high exposure between a lender and a borrower is the most important factor that determines whether they form a relationship.<sup>19</sup>

To this end, we employ several proxies of relationship lending that reflect the exclusivity of a relationship between a bank and a firm. First, we construct two measures considering the perspective of the borrower. *Main bank* is an indicator variable that equals 1 if the bank was the main lender of a given firm at the beginning of the semester (i.e., the largest fraction of a firm's debt comes from this bank), and 0 otherwise. We also use an alternative measure which is a refinement of the previous variable. *Main bank credit* > 50% is an indicator variable that equals 1 if the amount of credit outstanding with the main lender was more than 50% of the firm's total amount outstanding, and 0 otherwise. Second, we propose two additional measures considering the perspective of the lender: (i) *High exposure 90<sup>th</sup> percentile* is an indicator variable that equals 1 if a bank's exposure to a given firm (i.e., the amount of outstanding credit granted by the bank to the firm) is above the 90<sup>th</sup> percentile of the distribution of bank's exposures to

<sup>&</sup>lt;sup>18</sup> Our definition of the growth of credit follows Arce et al. (2021).

<sup>&</sup>lt;sup>19</sup> The exclusivity of a relationship can reflect both the intensity of the relationship in a given period and the creation of information over time. However, other variables, such as the frequency of interactions between a bank and a firm, may more accurately reflect the development of soft information (Kysucky and Norden, 2016). As described more fully below, inferences based on our main findings are unchanged when we use an alternative proxy for relationship lending, defined as the number of new interactions between a lender and borrower over the past five years.

individual firms. (ii) *High exposure 95<sup>th</sup> percentile* is defined similar to (i) but using the 95<sup>th</sup> percentile as the threshold.

We estimate the probability of default (PD) for all the firms in our sample for each year to assess if the introduction of the new provisioning model has a different effect on relationship lending for safe (investment grade) and risky (high yield) firms. The PD is based on the methodology developed in Blanco et al. (2023) for the Banco de España internal credit assessment. This measure follows –and extends– the approach of Altman (1968) to Spanish firms. Blanco et al. (2023) considers a firm to be in default if it has nonperforming loans for at least three months in a given year. We model the probability of default using a logit regression that includes five accounting ratios (own funds to total assets, financial expenditures to sales, ROA, liquid assets to total assets or gross value added to total assets) and the growth rate of aggregate credit to nonfinancial corporations. We estimate this model for six groups of firms based on their size (i.e., micro-firms vs. small, medium and large firms) and on their sector (i.e., manufacturing, construction and other sectors).<sup>20</sup>We use a firm's PD in 2015 (2017) as a control for a firm's credit risk in the pre-IFRS 9 (post-IFRS 9) period.

Table 1 provides the descriptive statistics for the main variables that we consider in our analysis, distinguishing between the period before and after IFRS 9. We first discuss our proxies of relationship lending. From the borrower perspective, approximately half of our bank-firm observations exhibit an exclusive relationship according to our first relationship lending variable, *Main Bank*. When considering the perspective of the lender (i.e., *High exposure 90<sup>th</sup> percentile* and *High exposure 95<sup>th</sup> percentile*), because we require lenders to have a very large exposure to a borrower (i.e., above the 90<sup>th</sup> and 95<sup>th</sup> percentile), the proportion of bank-firm pairs that form a relationship significantly drops. Our dependent variable, the biannual credit growth at the bank-firm level, exhibits a negative average value of approximately 24% during our period of analysis. The decrease is similar both before and after IFRS 9, which is indicative of a steady trend throughout this period, when the Spanish non-financial sector was still experiencing a deleveraging process.

Next, we consider the variables that are measured at the firm level. Focusing on the indicator variable that defines a high yield firm (i.e., PD>0.4%), we observe that, on average, 74% of the borrowers before and after 2018 fall under this category. This statistic is likely attributable to the high presence of SMEs in the Spanish corporate sector. Among firms that are defined as high yield, we observe a sizable fraction—17 % of borrowers in this subsample before 2018 and 22%

<sup>&</sup>lt;sup>20</sup> The category "Other sectors" comprises: Primary sector; Energy; Retail and wholesale trade; Hospitality, restoration and leisure; Transport and storage; Other market services; Motor vehicles.

after this date—experience an increase in their PD. In other words, a significant fraction of firms that are *ex-ante* risky experience a downgrade in their credit quality.

< Insert Table 1 here >

#### 4. Methodology: Identifying the effect of IFRS 9 on relationship lending

The main objective of this study is to examine the effect of IFRS-9 impairment model on relationship lending. As noted earlier, our identification faces several challenges. First, we need to restrict our sample to borrowers affected by the new provisioning methodology. In particular, the impact of the introduction of the ECL model is expected to be concentrated in borrowers with loans classified in Stage 1 and in Stage 2 and it should potentially be higher for Stage 2 loans. In contrast, the loan loss allowance associated with new credit to firms classified in Stage 3 should be similar under the ICL the ECL models. Hence, including these borrowers in our estimation likely would introduce a form of measurement error and bias in favor of finding no effect of IFRS 9 on relationship lending. Although the three-level asset classification did not exist prior to 2018 and is unobservable at the borrower level after the introduction of the new rule, as described in Section 2.2, we address these data limitations by classifying firms into different stages using observable information in our period of analysis (see Section 2.2). Our baseline regression considers borrowers that we classify in Stage 1, and we use firms defined as in Stage 3 to conduct a falsification test.

The second challenge we face is to disentangle the effect of IFRS 9 on relationship lending from credit supply trends during our period of analysis. Endogeneity bias could result if these factors were relevant determinants of the growth of credit and associated with the relationship strength between banks and firms. For example, a "low-for-long" interest rate environment, characterized by low interest rates over an extended period, has been shown to affect the provision of credit and risk taking (Arce et al., 2018). During our sample period, such an environment was present in the Eurozone, in part the result of European Central Bank (ECB) monetary policy. Also, the participation of banks in the non-regular open market operations of the ECB could affect the supply of credit for firms in our sample. We address the potential effects of these unobserved time-varying factors by introducing bank-time fixed effects, and by conducting cross-sectional tests to show that our effect becomes more relevant whenever we predict that IFRS 9 should have a higher impact.

We also control for the demand of credit by introducing firm-time fixed effects. Thus, we conduct our analysis on the sample of firms with multiple relationships and further show that our inferences are robust to the consideration of single-relation firms, employing industry-location-time fixed effects (Degryse et al., 2019). Finally, if banks changed their lending behavior prior to

the implementation of IFRS 9, our estimation would be biased in favor of finding no effect of IFRS 9 on relationship lending. We mitigate this possibility by conducting an anticipation test that shows no noticeable credit effects prior to implementation of IFRS 9.

We identify the effect of the ECL model on relationship banking by comparing the supply of credit of banks to their relationship borrowers, before and after the introduction of IFRS-9. In particular, we estimate the following regression equation at the bank-firm-semester level, focusing on borrowers with loans that we classify as being in Stage 1:

$$\Delta Credit_{fbt} = \beta_1 R L_{fbt-1} + \beta_2 R L_{fbt-1} \times Post_t + \alpha_{ft} + \gamma_{bt} + \epsilon_{fbt}$$
(1)

The dependent variable is the growth of credit between firm f and bank b in semester t. RL denotes our proxy of relationship banking, measured at the bank-firm-semester. We employ different definitions of this variable considering both the perspective of the lender and the borrower (see Section 3.2). Our main variable of interest is the interaction of RL and Post, a time indicator variable that equals zero for semesters before January 1, 2018 and one after this date. Thus,  $\beta_2$  reflects the incremental effect of RL on  $\Delta Credit$  associated with the ECL model. If the new provisioning methodology dilutes the role of relationship lending, we would expect to observe a negative coefficient on the interaction term,  $\beta_2$ . Our regression includes firm-time fixed effects,  $\alpha_{ft}$  and bank-time fixed effects,  $\gamma_{bt}$ . This regression model is similar to that of Sette and Gobbi (2015), which studies whether relationship lending mitigates the transmission of the Lehman default shock to the supply of credit in Italy.

#### 5. Stage 1 and relationship lending

#### 5.1 Baseline analysis

Table 2 presents the findings from the estimation of equation (1), pooling all firms that we classify as being in Stage 1. The Columns 1 and 2 (Columns 3 and 4) findings are based on estimations in which *RL* is measured from the perspective of the borrower (lender), *Main bank* and *Main bank credit* > 50% (*High exposure 90<sup>th</sup> percentile* and *High exposure 95<sup>th</sup> percentile*). The findings in Table 2 reveal a significantly negative effect of the new impairment model on the role of relationship lending across all four *RL* estimations.<sup>21</sup> The Column 1 findings indicate that a firm holding an exclusive relationship enjoys 10.3 percentage points (pp) higher semiannual credit growth before the implementation of IFRS 9, i.e, the *RL* coefficient = 0.103, but this effect falls by 1.8 pp

<sup>&</sup>lt;sup>21</sup> Untabulated findings from estimations that include firms for which we do not have enough information to estimate their PD before and after the introduction of IFRS and estimations in which we use log difference of credit as our dependent variable yield the same inferences as those based on the Table 2 findings. See Section 8 for additional tests assessing the robustness of inferences based on the baseline results.

following implementation. The Column 2 findings reveal the firms with more than 50% of the total amount outstanding have an even larger relative fall in credit growth, 2.1 pp, after 2018. The Columns 3 and 4 findings reveal a similar pattern, with the fall in credit growth being larger for borrowers in higher exposure percentiles. We observe a 1.9 pp and 2.5 pp decreases when a credit exposure is above the 90<sup>th</sup> and 95<sup>th</sup> percentiles of the distribution at the beginning of the semester.

The findings in Table 2 showing a positive effect of relationship lending both before and after 2018 (i.e., all of the *RL* and sum of *RL* and *RL* × *Past* coefficients are significantly positive), corroborate extant findings in the literature that relationship lending is a key that enables firms to access external finance (e.g., Sette and Gobbi, 2015; Bolton et al., 2016; Banerjee et al., 2021). However, the new impairment model appears to reduce the effect of relationship banking on credit growth. This finding suggests that IFRS 9 may reduce the incentives of lenders under an exclusive relationship to further increase their exposure to their relationship borrowers by granting new credit.

#### < Insert Table 2 here >

#### 5.2 Implications on the availability of credit

Equation (1) identifies the effect of IFRS 9 on a borrower's credit growth from a bank with a strong relationship compared to a bank with a weaker relationship. However, this specification does not address whether total lending to a borrower who relies more on relationship lending may have decreased as a result of IFRS 9. It is possible that these borrowers substituted the lending from banks with a strong relationship to banks with a weaker relationship.

To assess the impact of IFRS 9 on relationship lending at the firm level, we estimate a variation of equation (1), where the regression analysis is performed at the firm level instead of at the bank-firm level, and include firm-bank pairs with an existing relationship at the beginning of each semester.<sup>22</sup> We define credit growth as the change in a firm's total credit balance with all banks in one semester relative to the previous semester, normalized by the average credit balance in the two semesters. The explanatory variable of interest is *Share\_MainBank*, which we define as the fraction of the credit balance of a firm in its main bank at the beginning of each semester. To measure the differential effect of relationship lending when IFRS 9 is implemented, we interact *Share\_MainBank* with a time indicator, *Post*, that equals one after January 2018. To control for the effects of time-varying credit demand, we follow Cingano et al. (2016) and include the estimated firm-time-fixed effect ( $\alpha_{ft}$ ) from the bank-firm level regression (equation (1)), additional firm control variables, and industry-location-size-time fixed effects.

<sup>&</sup>lt;sup>22</sup> For this analysis, we also focus on firms that we classify as being in Stage 1.

Table 3 presents the findings. In Column 1, we measure *Share\_MainBank* as the amount of a firm's credit with its main bank relative to its total credit. In Column 2, we measure *Share\_MainBank* as the share of credit with the main bank only if it accounts for more than 50% of a firm's total credit. In both cases, the findings reveal that the interaction coefficients, –0.0243 and –0.0179, are significantly negative, which indicates that implementation of IFRS 9 reduces the positive effect of relationship lending on loan growth. These results are consistent with the evidence presented in Table 2 at the firmbank level. Furthermore, they suggest that firms that relied more on relationship lending experienced greater difficulties in accessing credit after the implementation of IFRS 9, and did not offset fully the loss of credit from relationship lenders by borrowing from non-relationship lenders.

#### < Insert Table 3 here >

To assess the economic relevance of the negative impact of IFRS 9 on relationship lending, we perform an approximate calculation. First, we calculate the reduction of credit for a borrower from its main bank after IFRS 9. To do so, we multiply the average outstanding credit of firms with their main bank at the end of the year 2017 by the annualized impact of IFRS 9 on relationship lending (i.e., the coefficient of *Share\_MainBank*  $\times$  *Post* in Column 1 of Table 3). Next, to approximate the aggregate reduction in credit we multiply this figure by the total number of non-financial corporations in Spain with positive outstanding credit at the end of the year 2017. Finally, we divide the aggregate reduction in credit by the total outstanding credit of Spanish non-financial corporations at the end of the year 2017. Our calculation suggests a sizeable effect on the availability of credit: the impact of IFRS 9 on relationship lending entailed an aggregate credit reduction of &13.4 billion in the year 2018, which represents 2.8% of the total outstanding credit of Spanish non-financial corporations at the end of 1018 at the end of 2017.

#### 5.3 Heterogenous effects at the bank level: The role of capital

An exclusive relationship implies a higher exposure between a bank and a firm compared to a less exclusive relationship. Our findings are consistent with IFRS 9 reducing the incentives of a bank to facilitate new credit to a borrower holding an exclusive relationship. The main intuition behind this result is that, after IFRS 9, banks may be less willing to further increase their exposure to borrowers for which they will have to recognize high loan loss provisions because provisions reduce regulatory capital (e.g., Abad and Suarez, 2018; Morais et al., 2020). Although we include controls in all our regressions for bank-time fixed effects, absorbing the direct effect of bank specific characteristics, it could be the case that the effect of IFRS 9 on relationship banking is more pronounced for banks that are less well-capitalized.

To test whether this is the case, we extend our baseline regression equation (1) by allowing for the effect of the new impairment model on relationship banking to differ for less well-capitalized banks. In particular, we include an indicator variable, *Low Capital*, that equals one if the bank's capital ratio is below the median in December 2015 (2017) for the pre-IFRS 9 (post-IFRS 9) period, and its interactions with *RL* and *RL* × *Low Capital*. The findings, presented in Table 4, reveal that each of the *RL* × *Low Capital* coefficients is significantly positive, ranging from 0.0766 to 0.0851. These findings indicate that prior to IFRS 9, the positive association between relationship lending and the growth of credit is larger for less well-capitalized banks. The findings are also consistent with the higher incentives of a less well-capitalized bank to facilitate new credit to an exclusive borrower to avoid impairment and thus, the recognition of loan losses.<sup>23</sup>

The findings reveal that each of the  $RL \times Low Capital \times Post$  coefficients on triple interaction, is negative but significantly so only marginally for borrowers with *High exposure 95<sup>th</sup> percentile*. Hence, there is, at best, weak evidence of a differential effect of IFRS 9 on relationship lending for less well-capitalized banks. Our results suggest that the change in the incentives for banks to strengthen an exclusive relationship is present for both good and bad capitalized banks.

< Insert Table 4 here >

#### 5.4 Heterogenous effects based on the role of loan guarantees

The expected loss given default of a loan is a key input to estimate loan loss provisions under IFRS 9. When a loan is secured by collateral, the shift from the ICL to the ECL model may have little implications for the loan loss provisions. The guarantees largely reduce the loss given default that the lender will face and, thus, expected credit losses. As a result, when considering new lending to a relationship borrower whose outstanding credit is guaranteed by collateral, banks may not have to foresee the need to recognize loan loss provisions at loan origination and throughout the life of the loan. In this case, banks may face similar incentives to facilitate new credit to an exclusive borrower before and after IFRS 9, and hence we should expect no change in the association between relationship banking and the growth of credit after IFRS 9.

To examine whether this is the case, we estimate equation (1) focusing on Stage 1 borrowers and including firm-bank pairs with guaranteed credit exposures at the beginning of each semester. The findings, presented in Table 5, reveal that each of the *RL* coefficients is significantly positive, ranging from 0.0110 to 0.0421. Thus, from the perspective of the borrower and the lender, holding an exclusive relationship has a positive effect on the growth of credit before implementation of IFRS 9. A comparison of the Table 2 and Table 5 *RL* coefficients

<sup>&</sup>lt;sup>23</sup> Banks may have incentives to provide new credit to a borrower to avoid impairment and the recognition of loan loss provisions, which would imply a deterioration of capital buffers. For borrowers very close to default, the literature refers to this practice as evergreening or zombie lending (e.g., Peek and Rosengren, 2005; Caballero et al., 2008). Because we consider the effect of IFRS 9 on relationship lending for firms in Stage 1, which have all their outstanding credit classified as performing, our study's findings are unlikely to reflect the effects of zombie lending.

reveals that the magnitude of the effect is higher in Table 2, which includes firm-bank pairs both with and without guaranteed credit. These findings are consistent with a large literature documenting the role of collateral as a tool to reduce information asymmetries (e.g., Stiglitz and Weiss, 1981; Bester, 1985) and suggest that relationship lending is less relevant in the presence of collateral (e.g., Holmstrom and Tirole, 1997).

More important, the Table 5 findings reveal that none of the  $RL \times Post$  coefficients is significant, which indicates that implementation of IFRS 9 had no impact on the effect of relationship banking on the growth of credit for collateralized borrowers. As expected, the new impairment model may have little impactions on loan loss provisions when considering borrowers with guaranteed credit exposures. In this case, IFRS 9 does not alter the incentives of relationship lending. More broadly, these results provide evidence that the impact of IFRS 9 relates to the lower willingness of banks to increase their exposure to borrowers when they foresee having to account for high expected credit losses.

< Insert Table 5 here >

#### 5.5 Heterogenous effects at the borrower level

#### 5.5.1 Ex-ante risk

Our primary findings in Table 2 indicate that there is a negative effect of the new impairment model on relationship lending. We next consider potential channels that help to explain this finding. In particular, we consider the role of borrower credit quality. The higher the *ex-ante* probability of default of a borrower (i.e., higher PD), the higher the estimated expected credit losses and the provisions for loan losses. Following the implementation of IFRS 9, banks may have reduced incentives to increase their exposure to *ex-ante* risky borrowers (e.g., Ertan, 2021; Morais et al., 2020; Bischof et al., 2022). Furthermore, this change in incentives may be larger when banks hold an exclusive relationship (i.e., the bank has a large share of the borrower's total credit). This is because, under the ICL model, banks had strong incentives to facilitate new lending to risky borrowers in exclusive relationships to avoid recognizing credit losses, since reducing lending to these borrowers could increase default risk (Boot, 2000). IFRS 9 reduces these incentives by requiring provisions for expected credit losses to be recognized at loan origination. In contrast, the new impairment model may not affect relationship lending to safe borrowers because expected credit losses for these firms are low. For this reason, we predict that the negative effect of IFRS 9 on relationship lending is stronger for firms with a riskier profile.

To test this prediction, we estimate a version of equation (1), partitioning the sample into investment grade (IG) and high yield firms (HY) (see section 2.2.1). The findings are presented in

Table 6. The key finding is that the RL x Post coefficients are significantly negative only for the HY estimations in Columns 2, 4, 6, and 8. The negative impact of IFRS 9 for HY firms ranges from a 2 pp decrease for the *Main bank* estimation to a 2.9 pp decrease for the *High exposure 95<sup>th</sup> percentile* estimation. In contrast, the effect of relationship lending on the growth of credit for investment grade borrowers is unchanged after IFRS 9. Hence, consistent with our prediction, the effect of the new impairment model on relationship lending is present in the sample of borrowers with a high *ex-ante* probability of default.

#### < Insert Table 6 here >

Overall, the effect of the new impairment model on relationship lending introduces a new asymmetry in the way a firm may benefit from holding an exclusive relationship to access external finance. Before 2018, relationship lending benefited investment-grade and high-yield firms in a similar way. However, after IFRS 9 was implemented, the effect of relationship lending on the supply of credit changed for the two types of borrowers. Our findings suggest that the higher cost of lending after IFRS 9 and the impact on relationship banking is borne mainly by high yield firms. From a policy standpoint, this is an important finding because riskier borrowers may face greater difficulty in accessing external finance when relationship banking could no longer be as effective in insulating their flow of credit over time, especially during recessions and liquidity shocks (e.g., Bolton et al., 2016; Banerjee et al., 2021).

#### 5.5.2 Credit quality deterioration

Next, we build on the comparison of investment grade and high yield firms in section 5.5.1 to identify two groups of firms that may be differentially affected by IFRS 9. Specifically, we consider the role of a deterioration in the credit quality of a borrower. We expect the impact of IFRS 9 on relationship lending to be stronger for riskier firms whose credit quality has deteriorated, as banks may anticipate higher loan loss provisions throughout the loan's life. This is because new loans to these borrowers are more likely to transition from Stage 1 to Stage 2, where lifetime PDs are considered, thereby resulting in a significant rise in provisions (ESRB, 2017).

To test our prediction, we estimate equation (1) for two subsamples: investment grade firms (IG) in Stage 1 with a stable credit quality (i.e., the PD has remained equal or below 0.4%) and high yield firms (HY) in Stage 1 whose credit quality has deteriorated (i.e., the PD has increased in the previous years). We define the change in credit quality of a firm in the pre-IFRS 9 period, i.e., before 2018, as the change in PD between 2014 and 2015. Similarly, we define the change in credit quality in the post-IFRS 9 period as the change in the PD between 2016 and 2017. Table 7, Panels A and B, presents the findings for each subsample.

The findings in Panel A reveal that none of the  $RL \times Post$  coefficients is significant. Hence, as expected, the new impairment model does not affect relationship banking for *ex-ante* low risk firms in Stage 1 whose credit risk has remained low. These findings suggest that the increase in the cost of credit that results from provisioning at origination under IFRS 9 will be low for such firms, and that banks perceive that a future transition between Stage 1 and 2 is unlikely to occur when granting new credit to this profile of borrowers.

The findings in Panel B, reveal that three of the four  $RL \times Post$  coefficients are significantly negative, with the significant coefficient ranging from -0.0214 to -0.0232. These findings indicate that the negative effect of the ECL on relationship lending is more pronounced for *ex-ante* high risk firms in Stage 1 that exhibit a deterioration in their credit quality and suggest that the outstanding credit for these borrowers is more likely migrate to Stage 2 in the future. In this case, banks may anticipate both larger provisions at loan origination and a higher probability that a transition between Stage 1 and Stage 2 occurs. Thus, after implementation of IFRS 9, banks are less willing to strengthen an exclusive relationship with borrowers that may more likely downgrade to Stage 2.

#### < Insert Table 7 here >

#### 6. The transition towards Stage 2

Supervisors provide some guidance to banks regarding what should be deemed a significant increase in credit risk, triggering a transition from Stage 1 to Stage 2. In particular, as noted above, they suggest that a threefold increase in the lifetime PD from initial recognition should be used as a rule of thumb. Although this is not a definitive rule that banks must follow, it enables us to explore the "cliff effect" under IFRS 9, which refers to the significant increase in loan loss provisions when a loan transitions from Stage 1 (12-month expected credit losses) to Stage 2 (lifetime expected credit losses). Under IFRS 9, new lending to borrowers with a significant deterioration in credit quality may imply provisioning for large expected credit losses not only at origination, but also if the new loan downgrades to Stage 2 in the future. Therefore, we estimate equation (1) for the sample of high yield firms that have experienced a threefold increase in their PDs. These are firms that were excluded in our previous analyses. Thus, we now focus our attention on borrowers whose outstanding credit may have very likely already transitioned to the Stage 2.

The results from these estimations presented in Table 8 provide support for the hypothesis that the negative impact of IFRS 9 on relationship lending is significantly accentuated when lenders anticipate a likely transition from the first to the second stage, as predicted by the "cliff-effect" feature of IFRS 9. In particular, the  $RL \times Post$  coefficient in the *Main bank credit* > 50 % estimation

indicates that there is a decrease the growth of credit by 5.7 pp. for relationship borrowers in the post-IFRS 9 period. This magnitude is twice as large as the negative impact of IFRS 9 on relationship lending for all Stage 1 firms with a high *ex-ante* credit risk shown in Table 6.

< Insert Table 8 here >

#### 7. Falsification test: Stage 3

The three-stage asset classification under IFRS 9 allows for the possibility of conducting a falsification test. In particular, we would not expect to observe a significant effect of IFRS 9 on relationship banking when the *ex-ante* credit risk of a borrower is sufficiently high, as reflected by the borrower's non-performing status. Under both the ICL model and IFRS 9, when extending credit to these borrowers, banks must recognize loan loss provisions based on credit losses that assume a PD close to 100% (e.g., Novotny-Farkas, 2016; López-Espinosa et al., 2021). Therefore, the loan loss allowance recorded on loans with objective evidence of impairment (i.e., Stage 3 loans) is likely to be similar under the ICL and ECL models.

Table 9 presents findings from estimations of equation (1) using a sample of borrowers whose outstanding credit may be allocated in Stage 3. As described in section 2.2.3, we classify a borrower as being in Stage 3 if all their outstanding credit is non-performing. This sample comprises credit impaired firms with a PD very close to 100%. This is a conservative definition of Stage 3 because a borrower could be potentially close to impairment (and effectively treated as credit impaired) and still have some credit with an under-performing or even performing status. The findings reveal that prior to the introduction of IFRS 9, holding a relationship entailed a growth of credit between 15.4 pp and 24.6 pp higher than the one observed for non-relationship borrowers. These results corroborate the significant role of relationship lending in extending credit to very risky borrowers that would, otherwise, have no access to external finance. More importantly, the findings reveal that none of the  $RL \times Post$  coefficients is significant, which indicates there is no significant differential impact of relationship lending on credit growth after the introduction of IFRS 9. This is consistent with the new regulation bringing no major changes regarding the provisioning methodology that is employed for credit impaired borrowers.

#### < Insert Table 9 here >

#### 8. Robustness tests and extensions of the baseline analysis

# 8.1 Single-relationship firms, anticipation effect, and alternative definition of relationship lending

In the previous analysis, we identified the effect of IFRS 9 on relationship lending controlling for the demand of credit with firm-time fixed effects. This approach focused on multi-bank firms, excluding those that borrow from only one bank. Although multi-relationship firms account for approximately 80% of the total loan volume in the Spanish economy, assessing whether our findings apply also to single-relationship firms is important to assess the external validity of our results. Accordingly, we estimate equation (1) using both single- and multi-bank firms. Following Degryse et al. (2019), we control for credit demand using industry-location-size-time fixed effects. The findings in Columns 1 and 2 of Table 10 reveal that IFRS 9 has a negative average impact on relationship lending, which is consistent with the findings relating to multi-bank firms and suggests that our findings can be generalized to both single- and multi-bank firms.

Next, we test for an anticipation effect, reflecting the possibility that banks may have changed their lending behavior before adopting IFRS 9. IFRS 9 was issued by the International Accounting Standards Board on July 2014 and was first applied in the financial statements for the 2018 fiscal year. Because of the significance of this regulatory change, financial institutions made substantial efforts to develop new provisioning methodologies and understand the consequences of the ECL model on their business models before the adoption of the new regulation. This suggests that banks may have changed their lending behavior, particularly the role of relationship lending, before adopting IFRS 9. If such an anticipation effect is present, our estimates of the effect of IFRS 9 on relationship lending likely represent a lower bound of the true effect of IFRS 9. To test for the possibility of an anticipation effect, we repeated our baseline regression estimation of equation (1), excluding all observations from the last semester of 2017 when the anticipation effect is likely to have been most prevalent. The findings in Column 3 and 4 of Table 10 reveal no evidence of an anticipation effect. In particular, the  $RL \times Post$  coefficients are similar to those in Table 2.

Finally, we assess whether our inferences are sensitive to use of an alternative definition of relationship lending that reflects the information a bank has about a firm. Our proxies for relationship lending focus on the exclusivity of a relationship from the perspective of the borrower and the lender. Thus, our proxy variables can reflect both the intensity of a relationship in a given period as well as the availability of information that facilitates loan contracting over time. Therefore, we consider an alternative definition of relationship lending that reflects primarily the information that a bank has about a firm, rather than focusing primarily on exposure.<sup>24</sup> Specifically, we develop a proxy for relationship lending as the number of new relationships between a lender and a borrower (i.e., the number of times that there is an increase in the outstanding amount of credit) over the past five years. We then estimate equation (1) using this alternative proxy of

<sup>&</sup>lt;sup>24</sup> The exclusivity of a relationship from the bank's perspective also can indicate credit concentration. An alternative interpretation of our results could be that IFRS 9 affects diversification incentives rather than relationship lending. This is unlikely for two reasons. First, we observe a negative effect of IFRS 9 using a measure of relationship lending that is not based on the exposure between a bank and a firm (Table 10). Second, the firm's perspective on exclusivity (i.e., a bank holding a large share of the borrower's total credit) is generally unrelated to the bank's diversification incentives.

relationship lending. Consistent with our main findings in Table 2, the findings in Column 5 of Table 10 reveal that IFRS 9 has a negative effect on relationship lending.

< Insert Table 10 here >

#### 8.2 The effect of IFRS 9 on relationship lending via lending rates

Our main analysis considers the effect of IFRS 9 on relationship lending focusing on the growth of credit. We document that IFRS 9 has a negative effect on relationship lending. It is plausible, however, that the effect of a higher cost of lending after IFRS 9 may translate into both a lower credit availability and higher lending rates for riskier borrowers under an exclusive relationship. We are not able to explore the impact of IFRS 9 on the cost of credit for our whole sample because lending rate data are only available since July 2018. Nonetheless, we conduct an exploratory empirical exercise considering a subsample of the main analysis.

Specifically, we examine the impact of IFRS 9 on the role of relationship lending via a change in lending rates by focusing on the credit granted before and after 2018 with a maturity greater than three years. This permits us to observe the interest rate of credit granted in the pre-adoption period (2016-2017) and in the post-adoption period (2018-2019).<sup>25</sup> Table 11 presents the results from estimating a version of equation (1) for the subsample using of loans with maturity greater than three years using the lending rate as the dependent variable. Consistent with the existing literature, we find that relationship lending facilitates access to credit as shown by the negative effect on interest rates (e.g., Petersen and Rajan, 1994; Elsas, 2005; Sette and Gobbi, 2015; Kysucky and Norden, 2016; López-Espinosa et al., 2017), i.e., all of the *RL* coefficients are significantly negative. However, we do not find that the effect of relationship lending on interest rates changes after the adoption of IFRS 9, i.e., none of the *RL* × *Post* coefficients is significant. These results suggest that, on average, the negative effect of IFRS 9 on relationship lending is manifested through the lower availability of credit rather than higher interest rates.

< Insert Table 11 here >

#### 9. Conclusion

A large literature provides theoretical and empirical evidence that relationship lending benefits firms and banks. An exclusive relationship between a bank and a firm is a key source of reduction of information frictions that can facilitate loan contracting over time. Also, when a bank holds a large share of the borrower's credit, there is a greater incentive for the bank to support the borrower by providing new credit in order to avoid impairment. However, there is little evidence whether and how a banking regulation affects relationship lending. Providing such evidence is

<sup>&</sup>lt;sup>25</sup> The main limitation of this strategy is that this subsample of loans may not be representative of the whole population, which limits the generalizability of the interest rate findings.

important to policy-makers and scholars alike given the central role of relationships in facilitating firms' access to external finance.

We fill this gap in the literature by focusing on the effect of the expected credit loss model under IFRS 9 on relationship lending. This is a major change in the way banks have to account for loan loss provisions, which are mandated to be forward-looking and raised at loan origination. Our results suggest that the new impairment model has reduced the role of relationship banking. In line with the existing literature, we show that holding an exclusive relationship is a relevant mechanism to enable firms to access external finance. Even though such a role remains after the introduction of IFRS 9, our study's findings suggest that the positive effect of relationship lending on the growth of credit is reduced. In addition, our results indicate that firms that relied more on relationship lending have experienced greater difficulties in accessing credit after the implementation of IFRS 9. The aggregate effect on credit availability is sizeable, with a reduction in credit to Spanish non-financial firms of 2.8% of their total outstanding credit. Finally, we document that the impact of IFRS 9 on relationship lending is present for the case of *ex-ante* riskier borrowers and whose credit quality has deteriorated. In contrast, the effect is significantly attenuated when considering new lending to a relationship borrower whose outstanding credit is guaranteed by collateral.

Our findings are consistent with a change in the incentives that underpin relationship lending. After IFRS 9, banks holding a more exclusive relationship with a firm are less willing to increase further their exposure if this results in larger expected credit losses at loan origination and a higher probability that the borrower downgrades to Stage 2 in future periods. Overall, our results suggest that regulatory changes may affect the role of relationship lending by affecting the "exclusivity dimension" of relationship banking. Specifically, banks may have less incentive to continue lending to exclusive relationship borrowers in order to avoid their default and the recognition of credit losses.

### References

- Abad, Jorge, and Javier Suarez. (2018). "The Procyclicality of Expected Credit Loss Provisions". Discussion Papers Series, 13135, CEPR. https://cepr.org/publications/dp13135
- Acharya, Viral V., and Stephen G. Ryan. (2016). "Banks' Financial Reporting and Financial System Stability". *Journal of Accounting Research*, 54(2), pp. 277-340. https://doi.org/10.1111/1475-679X.12114
- Altman, Edward I. (1968). "Financial Ratios, Discriminant Analysis and the Prediction of Corporate Bankruptcy". *The Journal of Finance*, 23(4), pp. 589-609. https://doi.org/10.2307/2978933
- Arce, Óscar, Miguel García-Posada, Sergio Mayordomo and Steven Ongena. (2018). "Bank Lending Policies and Monetary Policy: Some Lessons from the Negative Interest Era". *Economic Policy*, 38(116), pp. 899-941. https://doi.org/10.1093/epolic/eiae001
- Arce, Óscar, Sergio Mayordomo and Ricardo Gimeno. (2021). "Making Room for the Needy: The Credit-Reallocation Effects of the ECB's Corporate QE". *Review of Finance*, 25(1), pp. 43-84. https://doi.org/10.1093/rof/rfaa020
- Banerjee, Ryan N., Leonardo Gambacorta and Enrico Sette. (2021). "The Real Effects of Relationship Lending". *Journal of Financial Intermediation*, 48, pp. 1-14. https://doi.org/10.1016/j.jfi.2021.100923
- Beatty, Anne, and Scott Liao. (2011). "Do Delays in Expected Loss Recognition Affect Banks' Willingness to Lend?". Journal of Accounting and Economics, 52(1), pp. 1-20. https://doi.org/10.1016/j.jacceco.2011.02.002
- Berger, Allen N., and Gregory F. Udell. (1995). "Relationship Lending and Lines of Credit in Small Firm Finance". *The Journal of Business*, 68, pp. 351-381. https://doi.org/10.1086/296668
- Bester, Helmut. (1985). "Screening vs. Rationing in Credit Markets with Imperfect Information". *The American Economic Review*, 75(4), pp. 850-855. https://www.jstor.org/stable/1821362
- Bischof, Jannis, Rainer Haselmann, Frederik Kohl and Oliver Schlueter. (2022). "Limitations of Implementing an Expected Credit Loss Model". LawFin Working Papers, 48. https://doi.org/10.2139/ ssrn.4325220
- Bischof, Jannis, Christian Laux and Christian Leuz. (2021). "Accounting for Financial Stability: Bank Disclosure and Loss Recognition in the Financial Crisis". *Journal of Financial Economics*, 141(3), pp. 1188-1217. https://doi.org/10.1016/j.jfineco.2021.05.016
- Blanco, Roberto, Elena Fernández, Miguel García-Posada and Sergio Mayordomo. (2023). "An Estimation of the Default Probabilities of Spanish Non-Financial Corporations and Their Application to Evaluate Public Policies". Documentos Ocasionales, 2319, Banco de España. https://doi.org/10.53479/33512
- Bolton, Patrick, Xavier Freixas, Leonardo Gambacorta and Paolo E. Mistrulli. (2016). "Relationship and Transaction Lending in a Crisis". *Review of Financial Studies*, 29(10), pp. 2643-2676. https://doi.org/10.1093/ rfs/hhw041
- Boot, Arnoud W. A. (2000). "Relationship Banking: What Do We Know?". Journal of Financial Intermediation, 9(1), pp. 7-25. https://doi.org/10.1006/jfin.2000.0282
- Bushman, Robert M., and Christopher D. Williams. (2012). "Accounting Discretion, Loan Loss Provisioning, and Discipline of Banks' Risk-Taking". *Journal of Accounting and Economics*, 54(1), pp. 1-18. https://doi.org/10.1016/j.jacceco.2012.04.002

- Bushman, Robert M., and Christopher D. Williams. (2015). "Delayed Expected Loss Recognition and the Risk Profile of Banks". *Journal of Accounting Research*, 53(3), pp. 511-553. https://doi.org/10.1111/1475-679X.12079
- Caballero, Ricardo J., Takeo Hoshi and Anil K. Kashyap. (2008). "Zombie Lending and Depressed Restructuring in Japan". *American Economic Review*, 98(5), pp. 1943-1977. https://doi.org/10.1257/ aer.98.5.1943
- Cingano, Federico, Francesco Manaresi and Enrico Sette. (2016). "Does Credit Crunch Investment Down? New Evidence on the Real Effects of the Bank-Lending Channel". *Review of Financial Studies*, 29(10), pp. 2737-2773. https://doi.org/10.1093/rfs/hhw040
- Degryse, Hans, Olivier De Jonghe, Sanja Jakovljević, Klaas Mulier and Glenn Schepens. (2019). "Identifying Credit Supply Shocks with Bank-Firm Data: Methods and Applications". *Journal of Financial Intermediation*, 40, pp. 1-15. https://doi.org/10.1016/j.jfi.2019.01.004
- Degryse, Hans, Vasso Ioannidou and Steven Ongena. (2015). "Bank-Firm Relationships: A Review of the Implications for Firms and Banks in Normal and Crisis Times". In *The Economics of Interfirm Networks*, pp. 177-189. https://doi.org/10.1007/978-4-431-55390-8\_9
- De Haan, Leo, and Maarten R. C. van Oordt. (2018). "Timing of Banks' Loan Loss Provisioning during the Crisis". *Journal of Banking and Finance*, 87, pp. 293-303. https://doi.org/10.1016/j.jbankfin.2017.10.003
- Elsas, Ralf. (2005). "Empirical Determinants of Relationship Lending". *Journal of Financial Intermediation*, 14(1), pp. 32-57. https://doi.org/10.1016/j.jfi.2003.11.004
- Ertan, Aytekin. (2021). "Expected Losses, Unexpected Costs? Evidence from SME Credit Access under IFRS 9". Working Paper. https://papers.srn.com/sol3/papers.cfm?abstract\_id=3504708
- European Systemic Risk Board. (2017). "Financial Stability Implications of IFRS 9". https://www.esrb. europa.eu/pub/pdf/reports/20170717\_fin\_stab\_imp\_IFRS\_9.en.pdf
- Gaston, Ellen, and Inn W. Song. (2014). "Supervisory Roles in Loan Loss Provisioning in Countries Implementing IFRS". Working Papers, 2014/170, IMF. https://doi.org/10.5089/9781484381120.001
- Global Public Policy Committee. (2017). "The Auditor's Response to the Risks of Material Misstatement Posed by Estimates of Expected Credit Losses under IFRS 9: Considerations for the Audit Committees of Systemically Important Banks". https://www.iasplus.com/en/news/2017/07/ifrs-9impairment-banks-2
- Gropp, Reint, Thomas Mosk, Steven Ongena and Carlo Wix. (2019). "Banks Response to Higher Capital Requirements: Evidence from a Quasi-Natural Experiment". *Review of Financial Studies*, 32(1), pp. 266-299. https://doi.org/10.1093/rfs/hhy052
- Holmstrom, Beng, and Jean Tirole. (1997). "Financial Intermediation, Loanable Funds, and the Real Sector". *TheQuarterlyJournalofEconomics*, 112(3), pp.663-691. https://doi.org/10.1162/003355397555316
- Hu, Yunzhi, and Felipe Varas. (2021). "A Theory of Zombie Lending". *Journal of Finance*, 76(2021), pp. 1813-1867. https://doi.org/10.1111/jofi.13022
- Huizinga, Harry, and Luc Laeven. (2019). "The Procyclicality of Banking: Evidence from the Euro Area". *IMF Economic Review*, 67(3), pp. 496-527. https://doi.org/10.1057/s41308-019-00081-5

- Jiménez, Gabriel, Steven Ongena, José-Luis Peydró and Jesús Saurina. (2017). "Macroprudential Policy, Countercyclical Bank Capital Buffers, and Credit Supply: Evidence from the Spanish Dynamic Provisioning Experiments". *Journal of Political Economy*, 125(6), pp. 2126-2177. https://doi. org/10.1086/694289
- Khwaja, Asim I., and Atif Mian. (2008). "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market". American Economic Review, 98(4), pp. 1413-1442. https://doi.org/10.1257/ aer.98.4.1413
- Kysucky, Vlado, and Lars Norden. (2016). "The Benefits of Relationship Lending in a Cross-Country Context: A Meta-Analysis". *Management Science*, 62(1), pp. 90-110. https://doi.org/10.1287/ mnsc.2014.2088
- López-Espinosa, Germán, Sergio Mayordomo and Antonio Moreno. (2017). "When Does Relationship Lending Start to Pay?". Journal of Financial Intermediation, 31, pp. 16-29. https://doi.org/10.1016/j. jfi.2016.11.001
- López-Espinosa, Germán, Gaizka Ormazabal and Yuki Sakasai. (2021). "Switching From Incurred to Expected Loan Loss Provisioning: Early Evidence". *Journal of Accounting Research*, 59(3), pp. 757-804. https://doi.org/10.1111/1475-679X.12354
- Morais, Bernardo, Gaizka Ormazabal, Mónica Roa and Miguel Sarmiento. (2020). "Forward Looking Provisions: Credit Supply and Risk-Taking". Discussion Papers Series, 2020-027, CEPR. https://papers. ssrn.com/sol3/papers.cfm?abstract\_id=3702303
- Novotny-Farkas, Zoltán. (2016). "The Interaction of the IFRS9 Expected Loss Approach with Supervisory Rules and Implications for Financial Stability". *Accounting in Europe*, 13(2), pp. 197-227. https://doi. org/10.1080/17449480.2016.1210180
- Peek, Joe, and Eric S. Rosengren. (2005). "Unnatural Selection: Perverse Incentives and the Misalloation of Credit in Japan". American Economic Review, 95(4), pp. 1144-1166. https://doi.org/ 10.1257/0002828054825691
- Petersen, Mitchell A., and Raghuram G. Rajan. (1994). "The Benefits of Lending Relationships: Evidence from Small Business Data". *The Journal of Finance*, 49(1), pp. 3-37. https://doi.org/10.2307/2329133
- Sette, Enrico, and Giorgio Gobbi. (2015). "Relationship Lending During a Financial Crisis". *Journal of the European Economic Association*, 13(3), pp. 453-481. https://doi.org/10.1111/jeea.12111
- Stiglitz, Joseph E., and Andrew Weiss. (1981). "Credit Rationing in Markets with Imperfect Information". *The American Economic Review*, 71(3), pp. 393-410. https://www.jstor.org/stable/1802787

# FIGURE 1. EVOLUTION OF THE LOAN LOSS ALLOWANCE UNDER THE ICL AND ECL MODEL (IFRS 9)

This figure depicts the evolution of the loan loss allowance associated with a loan as its credit quality deteriorates. In the y-axis we represent the level of the allowance. In the x-axis we consider deterioration of credit quality since initial recognition (the increase in the PD of the loan). In red, we represent the scheme that results from the ICL model. We depict the evolution of the allowance according to the ECL model under IFRS 9 in blue. We further show the three-stage asset classification under IFRS 9 and the associated considered PD. Point A in the chart captures the initial allowance that is raised at origination under the ECL model. The particular level will depend on the PD and loss given default (LGD) of the borrower. We depict in the x-axis the significant increase in credit risk (SICR), the moment in where a loan classified in Stage 1 migrates to Stage 2.



# TABLE 1. DESCRIPTIVE STATISTICS ON RELATIONSHIP LENDING AND CREDIT RISK

Panels A and B of this table contain descriptive statistics (mean, standard deviation, median, 5th and 95th percentiles) on i) the relationship lending measures (main bank, main bank credit > 50 %, high exposure 90th percentile and high exposure 95th percentile), ii) the biannual credit growth at the bank-firm level, measured as the change in credit balance of firm f in bank b in a given semester with respect to the previous semester divided by the average credit balance in the two semesters, iii) ex-ante firms' credit risk (one-year estimated probability of default (PD) in December 2015 or 2017 depending on whether we use the pre- or post-even periods and the proportion of firms with PD>0.4%) and iv) ex-ante credit risk transition (firms with PD below or equal to 0.4% that had remained below this threshold in each period, firms with PD above 0.4% that had either increase less than three times or presented a threefold increase). We define the increase in PD based on the period 2014-2015 for the pre-event period and 2016-2017 for the post-event period. The sample only includes firm-bank pairs with an existing relationship at the beginning of each semester and it is restricted to firms with all their outstanding credit performing at the beginning of each semester. Additionally, for the case of the relationship lending measures, the biannual credit growth and exante firms' credit risk, the sample only includes firms which had not experienced a threefold-increase in their credit risk. Panel A contains the descriptive statistics for the pre-IFRS9 period, while Panel B refers to the post-IFRS9 period.

PANEL A	· PRF	IFRS	PERIOD	(ILINE 2016 -	DECEMBER	2017)
FAINELF	7. L UT.	-11.12.2	FERIOD	UUNE 2010 -	DECEMBER	201/)

Variable	Units	Mean	Std. Dev.	Median	5th percentile	95th percentile
Relationship lending						
Main bank	0/1	0.523	0.499	1	0	1
Main bank credit $> 50 \%$	0/1	0.474	0.499	0	0	1
High exposure 90th percentile	0/1	0.099	0.298	0	0	1
High exposure 95th percentile	0/1	0.045	0.207	0	0	0
Credit						
Biannual credit growth (bank-firm level)	-	-24.631	74.886	-11.403	-200.000	86.546
Credit risk						
Firm's probability of default	%	1.353	4.827	0.925	0.077	2.653
High yield firms (PD>0.4%)	0/1	0.777	0.416	1	0	1
Credit risk transition						
Safe firms (PD=<0.4%)	0/1	0.134	0.341	0	0	1
Risky firms(PD>0.4%) and:						
Increase in credit risk	0/1	0.168	0.374	0	0	1
Threefold increase in credit risk	0/1	0.029	0.168	0	0	0

#### PANEL B. POST-IFRS 9 PERIOD (DECEMBER 2017 - JUNE 2019)

Variable	Units	Mean	Std. Dev.	Median	5th percentile	95th percentile
Relationship lending						
Main bank	0/1	0.517	0.500	1	0	1
Main bank credit $> 50 \%$	0/1	0.469	0.499	0	0	1
High exposure 90th percentile	0/1	0.102	0.303	0	0	1
High exposure 95th percentile	0/1	0.047	0.211	0	0	0
Credit						
Biannual credit growth (bank-firm level)	-	-24.886	73.832	-12.290	-200.000	84.868
Credit risk						
Firm's probability of default	%	1.082	4.325	0.752	0.056	2.030
High yield firms (PD>0.4%)	0/1	0.713	0.452	1	0	1
Credit risk transition						
Safe firms (PD=<0.4%)	0/1	0.197	0.397	0	0	1
Risky firms(PD>0.4%) and:						
Increase in credit risk	0/1	0.221	0.415	0	0	1
Threefold increase in credit risk	0/1	0.033	0.178	0	0	0

### TABLE 2. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIAL REPORTING STANDARDS

This table reports the results obtained from the estimation of equation (1). The dependent variable is credit growth, which is measured as the change in credit balance of firm f in bank b in a given semester with respect to the previous semester divided by the average credit balance in the two semesters. Our first explanatory variable of interest is RL which is a dummy variable that takes the value of 1 if the bank and the firm had a lending relationship at the beginning of each semester. We employ four proxies of relationship lending. First, we use two measures considering the perspective of the borrower: (i) Main bank - in Column 1 - which is a dummy variable that takes the value of 1 if the bank was the main lender of a given firm at the beginning of the semester, and 0 otherwise, and (ii) Main bank credit > 50 % – in Column 2 - which is a refinement of the previous dummy variable such that it is equal to 1 if the amount of credit outstanding with the main lender was more than 50% of the total amount outstanding, and 0 otherwise. Second, we use other two measures considering the perspective of the lender: (i) High exposure 90th percentile- in Column 3- which is a dummy variable that takes the value of 1 if the firm was above the 90th percentile of the distribution of the amount of outstanding credit by firm and bank at the beginning of the semester and (ii) High exposure 95th percentile - in Column 4- for which we use the 95th percentile as the threshold. Our second explanatory variable of interest is the interaction of the measures of relationship lending and Post which is a dummy variable that takes the value of 1 for the period after the introduction of IFRS 9, i.e., from January 2018, and 0 before that date. All the regressions include firmtime and bank-time fixed effects. Our sample period spans from June 2016 to June 2019 and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. This is to guarantee that all the credit exposures of the banks in our sample to these firms can be classified in IFRS-9 stage 1. In addition, we restrict our sample to those firm-bank pairs with an existing relationship at the beginning of each semester. The row below the coefficients  $(RL + RL \times Post)$  contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

-	1 10xy J01 ILL					
	From the borr	ower's perspective	From the lender's perspective			
	Main bank	Main bank credit > 50 %		High exposure 95th percentile		
	(1)	(2)	(3)	(4)		
RL	0.1026***	0.1320***	0.0876***	0.0846***		
	(0.0041)	(0.0036)	(0.0087)	(0.0093)		
$RL \times Post$	-0.0177/*** (0.0037)	-0.0206*** (0.0042)	-0.0193*** (0.0058)	-0.0247*** (0.0060)		
$RL + RL \times Post$	0.0850***	0.111***	0.0683***	0.0599***		
	(0.00444)	(0.00395)	(0.00983)	(0.0101)		
Observations	2,269,993	2,269,993	2,269,993	2,269,993		
R-squared	0.3892	0.3899	0.3869	0.3865		
Firm × Time FE	Yes	Yes	Yes	Yes		
Bank × Time FE	Yes	Yes	Yes	Yes		

Proxy for RL

## TABLE 3. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIAL REPORTING STANDARDS. EVIDENCE AT THE FIRM LEVEL

This table reports the results obtained from the estimation of a variation of equation (1) in which regression analysis is conducted at the firm level instead of at the bank-firm level. Hence the dependent variable credit growth is measured as the change in the total credit balance of firm f in all banks in a given semester with respect to the previous semester divided by the average credit balance in the two semesters. The explanatory variable of interest is Share\_MainBank which is defined as the fraction of the credit balance of firm f in its main bank at the beginning of each semester and its interaction with a dummy variable that takes the value of 1 for the period after the introduction of IFRS 9, i.e., from January 2018, and 0 before that date. In Column 1, Share\_MainBank is defined as the share of credit of firm f in its main bank with respect to the total credit balance of firm f in all banks whereas in Column 2 it is the share of credit in the main bank only when it represents more than 50% of the total credit balance of firm f. We include credit demand controls in the form of the estimated firm fixed effect from the bank-firm-level regressions in equation (1), additional firm control variables (the logarithm of total assets, return on assets, leverage ratio, cash ratio and sales turnover ratio) and industry-location-size-time fixed effects. Only those firm-bank pairs with an existing relationship at the beginning of each semester are included and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefoldincrease in their credit risk. The row below the coefficients (Share\_MainBank + Share\_MainBank × Post) contains the linear combination of the two estimated coefficients of interest. Standard errors are clustered at the firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL				
	From the borrower's perspective				
	Main bank	Main bank credit > 50 %			
	(1)	(2)			
Share_MainBank	0.1074***	0.0734***			
Share_MainBank $\times$ Post	-0.0243*** (0.004)	-0.0179*** (0.002)			
Share_MainBank + Share_MainBank × Post	0.0831*** (0.0030)	0.0555***			
Observations	689,188	689,188			
R-squared Firm X Time FE	0.625160 No	0.625122 No			
Bank $\times$ Time FE	No	No			
$ILS \times Time FE$	Yes	Yes			
Credit Demand Control	Yes	Yes			
Firm Controls	Yes	Yes			

# TABLE 4. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIAL REPORTING STANDARDS. THE ROLE OF BANKS' CAPITAL RATIO

This table reports the results obtained from the estimation of an extended version of equation (1), in which additional explanatory variables of interest have been added to those defined in Table 2, namely the interaction of the measures of relationship lending and *Low capital* and the interaction of the measures of relationship lending and *Low capital* and the interaction of the measures of relationship lending and *Low capital* and the interaction of the measures of relationship lending, *Post* and *Low capital*. *Low capital* is a dummy variable that takes the value of 1 if the bank capital ratio is below the bottom quartile, and 0 otherwise. The dependent variable and the set of fixed effects are the same as those defined in Table 2. As in Table 2, only those firm-bank pairs with an existing relationship at the beginning of each semester are included and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL						
	From the perspec	ctive of the borrower	From the perspe	ctive of the lender			
	Main bank	Main bank credit > 50 %	High exposure 90th percentile	High exposure 95th percentile			
	(1)	(2)	(3)	(4)			
RL	0.0988***	0.1276***	0.0797***	0.0752***			
DI D	(0.0042)	(0.0040)	(0.0075)	(0.0074)			
KL × Post	$-0.01/2^{***}$	-0.0195***	-0.0138	-0.01/5*			
RL × Low capital	0.0781*	0.0851**	0.0766*	0.0846**			
$RL \times Low$ capital $\times Post$	(0.0469) -0.0202	(0.0388) -0.0293	(0.0445) -0.0556	(0.0419) -0.0661*			
	(0.0304)	(0.0297)	(0.0353)	(0.0358)			
Observations	2,269,993	2,269,993	2,269,993	2,269,993			
R-squared	0.3893	0.3899	0.3869	0.3866			
Firm × Time FE	Yes	Yes	Yes	Yes			
Bank × Time FE	Yes	Yes	Yes	Yes			

BANCO DE ESPAÑA **40 documento de trabajo n.º 2437** 

# TABLE 5. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIALREPORTING STANDARDS. THE CASE OF GUARANTEED CREDIT

This table reports the results obtained from the estimation of equation (1) as in Table 2, but with a different estimation sample given that it only includes guaranteed credit exposures. As in Table 2, we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. In addition, we restrict our sample to those firm-bank pairs with guaranteed credit exposures at the beginning of each semester. The dependent variable, the explanatory variables of interest and the set of fixed effects are similar to those defined in Table 2. The row below the coefficients ( $RL + RL \times Post$ ) contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL						
	From the perspec	tive of the borrower	From the perspective of the lender				
	Main bank	Main bank $Main bank credit > 50 \%$		High exposure 95th percentile			
	(1)	(2)	(3)	(4)			
RL	0.0110***	0.0193***	0.0295***	0.0421***			
	(0.0037)	(0.0040)	(0.0064)	(0.0074)			
$RL \times Post$	-0.0021	-0.0001	-0.0077	-0.0119			
	(0.0042)	(0.0051)	(0.0085)	(0.0108)			
$RL + RL \times Post$	0.00887*	0.0192***	0.0218***	0.0302***			
	(0.00492)	(0.00581)	(0.00807)	(0.00982)			
Observations	1,186,564	1,186,564	1,186,564	1,186,564			
R-squared	0.4067	0.4067	0.4067	0.4067			
Firm × Time FE	Yes	Yes	Yes	Yes			
$Bank \times Time FE$	Yes	Yes	Yes	Yes			

### TABLE 6. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIALREPORTING STANDARDS. THE ROLE OF EX-ANTE CREDIT RISK

This table reports the results obtained from the estimation of equation (1) for different subsamples of the firms used in Table 2 depending on their *ex-ante* levels of credit risk. The dependent variable, the explanatory variables of interest and the set of fixed effects are the same as those defined in Table 2. We split the firms in Table 2 in two groups based on their *ex-ante* probability of default (PD). Thus, investment grade (IG) firms are those with a one-year PD below or equal to 0.4%, which corresponds with CQS 1 – 3 according to the ECB classification, whereas high yield (HY) firms have a one-year PD above 0.4% and correspond to CQS 4- 8. The PD for the pre- and post-IFRS 9 periods are calculated with firms' balance-sheet information for 2015 and 2017, respectively. The row below the coefficients ( $RL + RL \times Post$ ) contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	F10xy J01 KL							
		From the borro	wer's perspecti	ve	From the lender's perspective			
	Mair	n bank Main credit >		Main bankHigh expcredit > 50 %90th perc		xposure High e ercentile 95th p		xposure ercentile
	IG	HY	IG	HY	IG	HY	IG	HY
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
RL	0.1165***	0.1000***	0.1412***	0.1302***	0.1049***	0.0850***	0.0832***	0.0843***
RL  imes Post	(0.0076) -0.0032	(0.0039) -0.0218***	(0.0072) -0.0045	(0.0038) -0.0254***	(0.0140) -0.0210*	(0.0085) -0.0195***	(0.0231) 0.0009	(0.0087) -0.0278***
	(0.0062)	(0.0037)	(0.0061)	(0.0046)	(0.0112)	(0.0060)	(0.0196)	(0.0065)
$RL + RL \times Post$	0.113*** (0.00848)	0.0782*** (0.00367)	0.137*** (0.00747)	0.105*** (0.00339)	0.0838*** (0.0146)	0.0655*** (0.00941)	0.0840*** (0.0160)	0.0565*** (0.00987)
	244.020	1 000 005	244.020	1 000 005	244.020	1 002 005	244.020	1 0 0 0 0 0 5
Observations	366,039	1,903,805	366,039	1,903,805	366,039	1,903,805	366,039	1,903,805
R-squared	0.4311	0.3791	0.4318	0.3797	0.4278	0.3769	0.4275	0.3766
Firm×Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Bank×Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Proxy for RL

# TABLE 7. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIALREPORTING STANDARDS. THE ROLE OF THE CHANGE IN CREDIT RISK

This table reports the results obtained from the estimation of equation (1) based on different subsamples of the firms used in Table 2. The group of firms considered in Panel A (from those of Table 2) had a probability of default (PD) below or equal to 0.4% in 2014 (for the pre-event period) and in 2016 (for the post-event period) and this PD had remained equal or below 0.4% one year later (i.e., in 2015 and 2017, respectively). The group of firms considered in Panel B (from those of Table 2) had a probability of default (PD) above 0.4% in 2015 (for the pre-event period) and in 2017 (for the post-event period) and this PD had increased with respect to one year before (i.e., in 2014 and 2016, respectively). The dependent variable, the explanatory variables of interest and the set of fixed effects are the same as those defined in Table 2. The row below the coefficients ( $RL + RL \times Post$ ) contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for KL					
	From the perspec	tive of the borrower	From the perspective of the lender			
	Main bank	Main bank credit > 50 %	High exposure 90th percentile	High exposure 95th percentile		
	(1)	(2)	(3)	(4)		
RL	0.1248***	0.1512***	0.1074***	0.0851***		
	(0.0099)	(0.0095)	(0.0156)	(0.0279)		
$RL \times Post$	-0.0027	-0.0063	-0.0170	-0.0022		
	(0.0075)	(0.0085)	(0.0155)	(0.0286)		
$RL + RL \times Post$	0.122***	0.145***	0.0904***	0.0829***		
	(0.00865)	(0.00779)	(0.0171)	(0.0186)		
Observations	227,085	227,085	227,085	227,085		
R-squared	0.4397	0.4405	0.4358	0.4354		
Firm × Time FE	Yes	Yes	Yes	Yes		
Bank × Time FE	Yes	Yes	Yes	Yes		

#### Panel A: Ex-ante safe firms (PD<0.4%) whose credit has remained low

Panel B: Ex-ante risky firms (PD>0.4%) which have experienced an increase in their PDs

	Proxy for KL				
	From the perspec	tive of the borrower	From the perspective of the lender		
	Main bank	Main bank credit > 50 %	High exposure 90th percentile	High exposure 95th percentile	
	(1)	(2)	(3)	(4)	
RL	0.0942***	0.1218***	0.0747***	0.0774***	
	(0.0047)	(0.0050)	(0.0107)	(0.0128)	
$RL \times Post$	-0.0214***	-0.0232***	-0.0114	-0.0222*	
	(0.0044)	(0.0052)	(0.0085)	(0.0128)	
$RL + RL \times Post$	0.0727***	0.0986***	0.0633***	0.0552***	
	(0.00473)	(0.00473)	(0.0115)	(0.0131)	
Observations	499,669	499,669	499,669	499,669	
R-squared	0.3919	0.3925	0.3902	0.3899	
Firm × Time FE	Yes	Yes	Yes	Yes	
Bank × Time FE	Yes	Yes	Yes	Yes	

#### TABLE 8. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIAL REPORTING STANDARDS. THE CASE OF EX-ANTE RISKY FIRMS THAT HAD EXPERIENCED A THREEFOLD INCREASE IN CREDIR RISK

This table reports the results obtained from the estimation of equation (1) based on a sample of firms which had aprobability of default (PD) above 0.4% in 2015 (for the pre-event period) and in 2017 (for the postevent period) and this PD had experienced a threefold increase with respect to one year before (i.e., in 2014 and 2016, respectively). The dependent variable, the explanatory variables of interest and the set of fixed effects are similar to those defined in Table 2. The row below the coefficients ( $RL + RL \times Post$ ) contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL						
	From the perspec	tive of the borrower	From the perspective of the lender				
	Main bank	Main bank credit > 50 %	High exposure 90th percentile	High exposure 95th percentile			
	(5)	(6)	(7)	(8)			
RL	0.1138***	0.1371***	0.1432***	0.1089***			
	(0.0147)	(0.0149)	(0.0319)	(0.0334)			
$RL \times Post$	-0.0449**	-0.0564***	-0.0751**	-0.0863*			
	(0.0198)	(0.0206)	(0.0352)	(0.0453)			
$RL + RL \times Post$	0.0690***	0.0807***	0.0680***	0.0225			
	(0.0111)	(0.0127)	(0.0258)	(0.0357)			
	10.000	40.000	12.222	10.000			
Observations	43,333	43,333	43,333	43,333			
R-squared	0.4595	0.4600	0.4574	0.4568			
Firm × Time FE	Yes	Yes	Yes	Yes			
Bank × Time FE	Yes	Yes	Yes	Yes			

## TABLE 9. FALSIFICATION TEST: RELATIONSHIP LENDING AND STAGE 3EXPOSURES

This table reports the results obtained from the estimation of equation (1) based on firms that are supposed to be in IFRS-9 stage 3 instead of stage 1 as in Table 2. We consider that a firm is in stage 3 when all its loans have a non-performing status. The dependent variable, the explanatory variables of interest and the set of fixed effects are the same as those defined in Table 2. The row below the coefficients ( $RL + RL \times Post$ ) contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL					
	From the perspec	tive of the borrower	From the perspective of the lender			
	Main bank	Main bank credit > 50 %		High exposure 95th percentile		
	(1)	(2)	(3)	(4)		
RL	0.2339***	0.2462***	0.1543***	0.1841***		
	(0.0231)	(0.0250)	(0.0561)	(0.0551)		
$RL \times Post$	-0.0362	-0.0164	-0.1069	-0.1568		
	(0.0374)	(0.0413)	(0.1031)	(0.1055)		
$RL + RL \times Post$	0.198***	0.230***	0.0474	0.0273		
	(0.0348)	(0.0383)	(0.0753)	(0.0951)		
Observations	7,266	7,266	7,266	7,266		
R-squared	0.5571	0.5580	0.5431	0.5429		
Firm × Time FE	Yes	Yes	Yes	Yes		
Bank × Time FE	Yes	Yes	Yes	Yes		

#### TABLE 10. ROBUSTNESS TESTS. SINGLE-RELATIONSHIP FIRMS, ANTICIPATION EFFECT AND ALTERNATIVE DEFINITION OF REALATIONSHIP LENDING

This table presents three robustness tests to our baseline result (Table 2). Columns 1 and 2 report the results obtained from the estimation of equation (1) as in Table 2 but including both single- and multi-bank firms. The dependent variable and the explanatory variables of interest are those defined in Table 2 and banktime fixed effects are also included, but we control for the demand of credit using industry-location-sizetime fixed effects. In addition, we include firm control variables (the logarithm of total assets, return on assets, leverage ratio, cash ratio and sales turnover ratio). As in Table 2, only those firm-bank pairs with an existing relationship at the beginning of each semester are included and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. Columns 3 and 4 report the results obtained from the estimation of equation (1) as in Table 2 but excluding all observations that belong to the last semester of the year 2017, when the anticipation effect may have been most prevalent. The dependent variable, the explanatory variables of interest and the set of fixed effects are the same as those defined in Table 2. As in Table 2, only those firm-bank pairs with an existing relationship at the beginning of each semester are included and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. Column 5 reports the results obtained from the estimation of equation (1) as in Table 2 but considering an alternative definition of relationship lending which captures the information that a given bank has about a given firm rather than the exposure. In particular, relationship lending is proxied as the number of new relationships between a lender and a borrower (i.e., the number of times that there is an increase in the outstanding amount of credit) during the last five years. The dependent variable and the set of fixed effects are similar to those defined in Table 2. In all columns, the row below the coefficients  $(RL + RL \times Post)$  contains the linear combination of the two estimated coefficients. Standard errors are clustered at the bank and firm level and are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Kooustness Tests									
	Single-rela	tionship firms	Anticip	Alternative definition of relationship lending						
	Main bank	Main bank credit > 50 %	Main bank	Main bank credit > 50 %	Number of new relationships during the last five years					
	(1)	(2)	(3)	(4)	(5)					
RL	0.0324*** (0.0065)	0.0198*** (0.0050)	0.1079*** (0.0044)	0.1381*** (0.0042)	0.0061*** (0.0003)					
$RL \times Post$	-0.0099** (0.0046)	-0.0083** (0.0039)	-0.0229*** (0.0039)	-0.0266*** (0.0045)	-0.0005** (0.0002)					
$RL + RL \times Post$	0.0225*** (0.00447)	0.0115*** (0.00381)	0.0850*** (0.00444)	0.111*** (0.00395)	0.0056*** (0.0002)					
Observations R-squared	3,208,516 0.0599	3,208,516 0.0597	1,883,994 0.3890	1,883,994 0.3897	2,001,982 0,3876					
Firm × Time FE	No	No	Yes	Yes	Yes					
Bank × Time FE	Yes	Yes	Yes	Yes	Yes					
ILS $\times$ Time FE	Yes	Yes	No	No	No					
Firm Controls	Yes	Yes	No	No	No					

D 1 -

### TABLE 11. RELATIONSHIP LENDING AFTER THE CHANGE IN FINANCIALREPORTING STANDARDS. THE EFFECT VIA LENDING RATES

This table reports the results obtained from the estimation of an alternative version of equation (1), in which the dependent variable is the lending interest rate at the loan level for a small subsample of the firms used in Table 2. As in Table 2, only those firm-bank pairs with an existing relationship at the beginning of each semester are included and we restrict our sample to firms with all their outstanding credit performing at the beginning of each semester and which had not experienced a threefold-increase in their credit risk. In addition, we only consider loans with an original maturity over 3 years. The explanatory variables of interest are those defined in Table 2. Firm-time fixed effects are included in Columns 1, 3, 5 and 7, whereas in Columns 2, 4, 6 and 8 we control for the demand of credit using industry-location-size-time fixed effects and include a set of control variables to account for firm specific characteristics (the logarithm of total assets, return on assets, leverage ratio, cash ratio and sales turnover ratio), loan amount, and maturity. In addition, all the regressions include bank-time and product-time fixed effects (with the later to control for loan type). Standard errors clustered at the bank-firm level are reported in brackets. \*, \*\* and \*\*\* indicate statistical significance at the 10%, 5% and 1% levels, respectively.

	Proxy for RL										
_	From the borrower's perspective				From the lender's perspective						
	Main bank		Main bank credit > 50 %		High exposure 90th percentile		High exposure 95th percentile				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)			
RL	-0.0547*	-0.0538*	-0.1782***	-0.0622*	-0.0087	-0.2531***	-0.1643***	-0.2641***			
$RL \times Post$	(0.0278) -0.0412	(0.0311) -0.0535	(0.0480) 0.0177	(0.0325) -0.0434	(0.0426) -0.0834	(0.0632) -0.0411	(0.0428) 0.0698	(0.0541) 0.0614			
	(0.0493)	(0.0426)	(0.0646)	(0.0466)	(0.0529)	(0.0622)	(0.0530)	(0.0558)			
Observations	30,099	91,950	30,099	91,950	30,099	91,950	30,099	91,950			
R-squared	0.9292	0.7201	0.9293	0.7201	0.9291	0.7210	0.9292	0.7203			
Firm × Time FE	Yes	No	Yes	No	Yes	No	Yes	No			
Bank × Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes			
$ILS \times Time FE$	No	Yes	No	Yes	No	Yes	No	Yes			
$Product \times Time FE$	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes			
Firm Controls	No	Yes	No	Yes	No	Yes	No	Yes			

### Appendix - Detailed explanation of the ICL and ECL model

### A1. The Incurred Credit Loss model

Prior to January 2018, banks following international accounting standards estimated loan loss provisions based on the Incurred Credit Loss model, as dictated by IAS 39. The ICL model required banks to materialize the accrual expense when there was objective evidence that impairment had occurred, constraining the incorporation of forward-looking information in the estimation of loan loss provisions. Hence, even if a bank could anticipate future events affecting the probability of default of a loan, early recognition was limited by the accounting rule. Impairment recognition had to be triggered by one or more 'loss events', such as a default or delinquency in interest or principal payments, indicating verifiable evidence of probable losses, as of the balance-sheet date. In practice, this entailed that loan losses were mainly considered when the probability of default (PD) was close to 100% (Novotny-Farkas, 2016).

While this methodology responded to the conservatism principle that is prevalent in many accounting practices, the 'backward-looking' nature of the ICL model was argued to have detrimental effects on financial stability, as witnessed during the financial crisis (Gaston and Song, 2014; Haan, de and van Oordt, 2018). The untimely and insufficient provisions that followed from the ICL model were blamed to exacerbate the severity of the crisis. Namely, delayed impairment recognition led to corrective provisions in the middle of the recession, increasing banks' capital inadequacy concerns and accentuating the ongoing credit crunch in the economy (Acharya and Ryan, 2016; Huizinga and Laeven, 2019; Bischof et al., 2021).

In the aftermath of the financial crisis, in response to this concern, the G-20 urged accounting setters to review the ICL methodology. Credit loss recognition was demanded to incorporate a more forward-looking approach. To this end, the International Accounting Standard Board replaced IAS 39 by IFRS 9, which was issued in July 2014 and effectively adopted for the first time in the year 2018. In a similar vein, the Federal Accounting Standard Board in the US introduced ASU 2016-13, adopted by listed banks in January 2020. In both versions, provisions were required to follow the Expected Credit Loss model, which is forward-looking in nature. The introduction of IFRS 9 in Spain in January 2018 through Circular 4/2017 represents the focus of our paper.

### A2. The Expected Credit Loss model

The transition from the ICL model to a provisioning approach based on expected credit losses represents a very significant switch in the way banks must provision for loan losses. Under the Expected Credit Loss model, loan loss provisions are forward-looking and raised at origination or purchase of a loan. After initial recognition, entities must update the expected credit loss associated to the financial asset at each reporting date, considering all reasonable and supportable information, including forward-looking information. In general terms, the estimation of the expected credit loss of a loan is based on the following formula:

#### $Expected \ Loss = PD \times LGD \times EAD$

where PD is the expected probability of default of the loan, LGD denotes the expected loss given default and EAD refers to the exposure at default (the outstanding debt at the time of default). The estimates of PD and LGD are scenario based. Namely, these inputs are computed under different macroeconomic scenarios using forecasts of key determinants such as, for example, GDP growth, unemployment rate or housing prices and typically considering a 'baseline' scenario as well as optimistic and pessimistic deviations. The final PD and LGD that enters in the expected loss

formula results from weighting these scenario-based inputs with the probability that each scenario occurs. While this general description is common to both IFRS 9 and ASU 16-13 in the US, the two standards significantly differ in the considered time horizons for the estimation of the PDs. Notably, in the former approach the time horizon depends on the performing status of the loans whereas full-life PDs are applied for all financial instruments in the latter methodology, no matter their credit quality.

### A3. Three-stage asset classification under IFRS 9

Under IFRS 9, the estimation of expected credit losses is based on a 3-stage asset classification. Loans are allocated into three different buckets according to their credit quality deterioration, considering different time horizons in each bucket to estimate the PD of the financial instruments. In particular, the expected credit loss of assets classified in stage 1 is based on 12-months PDs (i.e., the lender estimates the probability that the client will default in the next 12 months). Newly issued non-impaired loans and loans that have not experienced a significant increase in credit risk since origination are classified in this first bucket.

When a loan suffers a significant deterioration in credit quality, the asset is transferred into stage 2. In this second stage, banks must estimate the PD based on the remaining lifetime of the loan (full-life PDs). For this transition to be considered, IFRS 9 provides a rebuttable presumption that the credit risk of a financial asset has increased significantly since initial recognition when contractual payments are more than 30 days past due. The standard, however, requires banks to complement this criterion with both qualitative and quantitative information. Finally, if the credit risk of a financial asset rises to the point where it is regarded as credit impaired (PD=100%), the loan is classified in stage 3, in which lifetime expected credit losses are recognized.<sup>26</sup>

Hence, the crucial difference and impact of the new standard relative to the ICL model stems from the classification of exposures under stage 1 and stage 2, inexistent in the previous methodology. Notably, IAS 39 already considered raising provisions for loans with objective evidence of impairment at the reporting date (i.e., the loans that are classified in stage 3 under IFRS 9) and both methodologies were largely consistent in the treatment of purchased or originated credit-impaired assets. Under IFRS 9, provisions are raised at loans' origination, which are initially classified in stage 1. In sharp contrast to the ICL model, the credit risk assumed by a bank when granting a new loan will have immediate implications on provisions, earnings and regulatory capital, increasing the cost of risk that a bank incurs in his lending decision. Furthermore, provisions will have to be updated when the credit risk of the borrower changes significantly since initial recognition. This means that banks' lending decisions may be affected by both the initial raise in provisions at origination and the expectation of future corrections in the allowance. Such corrections may be especially relevant if a loan migrates from stage 1 to stage 2 as banks must start considering life-time PDs instead of 12-months PDs (ESRB, 2017).

<sup>&</sup>lt;sup>26</sup> In addition, for loans classified in stage 3, the amortized cost net carrying amount is used to compute interest revenue (i.e., reduced for expected credit losses). In contrast, for loans classified in stage 1 and stage 2, interest revenue is still calculated on the gross carrying amount of the asset.

#### **BANCO DE ESPAÑA PUBLICATIONS**

#### WORKING PAPERS

- 2320 JUAN S. MORA-SANGUINETTI, LAURA HOSPIDO and ANDRÉS ATIENZA-MAESO: The numbers of equality regulation. Quantifying regulatory activity on non-discrimination and its relationship with gender gaps in the labour market.
- 2321 ANDRES ALONSO-ROBISCO and JOSÉ MANUEL CARBÓ: Analysis of CBDC Narrative of Central Banks using Large Language Models.
- 2322 STEFANIA ALBANESI, ANTÓNIO DIAS DA SILVA, JUAN F. JIMENO, ANA LAMO and ALENA WABITSCH: New technologies and jobs in Europe.
- 2323 JOSÉ E. GUTIÉRREZ: Optimal regulation of credit lines.
- 2324 MERCEDES DE LUIS, EMILIO RODRÍGUEZ and DIEGO TORRES: Machine learning applied to active fixed-income portfolio management: a Lasso logit approach.
- 2325 SELVA BAHAR BAZIKI, MARÍA J. NIETO and RIMA TURK-ARISS: Sovereign portfolio composition and bank risk: the case of European banks.
- 2326 ANGEL-IVAN MORENO and TERESA CAMINERO: Assessing the data challenges of climate-related disclosures in european banks. A text mining study.
- 2327 JULIO GÁLVEZ: Household portfolio choices under (non-)linear income risk: an empirical framework.
- 2328 NATASCHA HINTERLANG: Effects of Carbon Pricing in Germany and Spain: An Assessment with EMuSe.
- 2329 RODOLFO CAMPOS, SAMUEL PIENKNAGURA and JACOPO TIMINI: How far has globalization gone? A tale of two regions.
- 2330 NICOLÁS FORTEZA and SANDRA GARCÍA-URIBE: A Score Function to Prioritize Editing in Household Survey Data: A Machine Learning Approach.
- 2331 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Taxing consumption in unequal economies.
- 2332 ESTHER CÁCERES and MATÍAS LAMAS: Dividend Restrictions and Search for Income.
- 2333 MARGARITA MACHELETT: Gender price gaps and competition: Evidence from a correspondence study.
- 2334 ANTON NAKOV and CARLOS THOMAS: Climate-conscious monetary policy.
- 2335 RICARDO BARAHONA, STEFANO CASSELLA and KRISTY A. E. JANSEN: Do teams alleviate or exacerbate the extrapolation bias in the stock market?
- 2336 JUAN S. MORA-SANGUINETTI and ANDRÉS ATIENZA-MAESO: "Green regulation": A quantification of regulations related to renewable energy, sustainable transport, pollution and energy efficiency between 2000 and 2022.
- 2401 LAURA HOSPIDO, NAGORE IRIBERRI and MARGARITA MACHELETT: Gender gaps in financial literacy: a multi-arm RCT to break the response bias in surveys.
- 2402 RUBÉN DOMÍNGUEZ-DÍAZ, SAMUEL HURTADO and CAROLINA MENÉNDEZ: The medium-term effects of investment stimulus.
- 2403 CLODOMIRO FERREIRA, JOSÉ MIGUEL LEIVA, GALO NUÑO, ÁLVARO ORTIZ, TOMASA RODRIGO and SIRENIA VAZQUEZ: The heterogeneous impact of inflation on households' balance sheets.
- 2404 JORGE ABAD, GALO NUÑO and CARLOS THOMAS: CBDC and the operational framework of monetary policy.
- 2405 STÉPHANE BONHOMME and ANGELA DENIS: Estimating individual responses when tomorrow matters.
- 2406 LAURA ÁLVAREZ-ROMÁN, SERGIO MAYORDOMO, CARLES VERGARA-ALERT and XAVIER VIVES: Climate risk, soft information and credit supply.
- 2407 JESÚS FERNÁNDEZ-VILLAVERDE, JOËL MARBET, GALO NUÑO and OMAR RACHEDI: Inequality and the zero lower bound.
- 2408 PABLO BURRIEL, MAR DELGADO-TÉLLEZ, CAMILA FIGUEROA, IVÁN KATARYNIUK and JAVIER J. PÉREZ: Estimating the contribution of macroeconomic factors to sovereign bond spreads in the euro area.
- 2409 LUIS E. ROJAS and DOMINIK THALER: The bright side of the doom loop: banks' sovereign exposure and default incentives.
- 2410 SALOMÓN GARCÍA-VILLEGAS and ENRIC MARTORELL: Climate transition risk and the role of bank capital requirements.
- 2411 MIKEL BEDAYO and JORGE E. GALÁN: The impact of the Countercyclical Capital Buffer on credit: Evidence from its accumulation and release before and during COVID-19.
- 2412 EFFROSYNI ADAMOPOULOU, LUIS DÍEZ-CATALÁN and ERNESTO VILLANUEVA: Staggered contracts and unemployment during recessions.
- 2413 LUIS FÉRNANDEZ LAFUERZA and JORGE E. GALÁN: Should macroprudential policy target corporate lending? Evidence from credit standards and defaults.
- 2414 STÉPHANE BONHOMME and ANGELA DENIS: Estimating heterogeneous effects: applications to labor economics.

- 2415 LUIS GUIROLA, LAURA HOSPIDO and ANDREA WEBER: Family and career: An analysis across Europe and North America.
- 2416 GERALD P. DWYER, BILJANA GILEVSKA, MARÍA J. NIETO and MARGARITA SAMARTÍN: The effects of the ECB's unconventional monetary policies from 2011 to 2018 on banking assets.
- 2417 NICOLÁS FORTEZA, ELVIRA PRADES and MARC ROCA: Analysing the VAT cut pass-through in Spain using webscraped supermarket data and machine learning.
- 2418 JOSÉ-ELÍAS GALLEGOS: HANK beyond FIRE: Amplification, forward guidance, and belief shocks.
- 2419 DANIEL ALONSO: Stabilisation properties of a SURE-like European unemployment insurance.
- 2420 FRANCISCO GONZÁLEZ, JOSÉ E. GUTIÉRREZ and JOSÉ MARÍA SERENA: Shadow seniority? Lending relationships and borrowers' selective default.
- 2421 ROBERTO BLANCO, MIGUEL GARCÍA-POSADA, SERGIO MAYORDOMO and MARÍA RODRÍGUEZ-MORENO: Access to credit and firm survival during a crisis: the case of zero-bank-debt firms.
- 2422 FERNANDO CEREZO, PABLO GIRÓN, MARÍA T. GONZÁLEZ-PÉREZ and ROBERTO PASCUAL: The impact of sovereign debt purchase programmes. A case study: the Spanish-to-Portuguese bond yield spread.
- 2423 EDGAR SILGADO-GÓMEZ: Sovereign uncertainty.
- 2424 CLODOMIRO FERREIRA, JULIO GÁLVEZ and MYROSLAV PIDKUYKO: Housing tenure, consumption and household debt: life-cycle dynamics during a housing bust in Spain.
- 2425 RUBÉN DOMÍNGUEZ-DÍAZ and SAMUEL HURTADO: Green energy transition and vulnerability to external shocks.
- 2426 JOSEP GISBERT and JOSÉ E. GUTIÉRREZ: Bridging the gap? Fintech and financial inclusion.
- 2427 RODOLFO G. CAMPOS, MARIO LARCH, JACOPO TIMINI, ELENA VIDAL and YOTO V. YOTOV: Does the WTO Promote Trade? A Meta-analysis.
- 2428 SONER BASKAYA, JOSÉ E. GUTIÉRREZ, JOSÉ MARÍA SERENA and SERAFEIM TSOUKAS: Bank supervision and non-performing loan cleansing.
- 2429 TODD E. CLARK, GERGELY GANICS, and ELMAR MERTENS: Constructing fan charts from the ragged edge of SPF forecasts.
- 2430 MIGUEL GARCÍA-POSADA and PETER PAZ: The transmission of monetary policy to credit supply in the euro area.
- 2431 KLODIANA ISTREFI, FLORENS ODENDAHL and GIULIA SESTIERI: ECB communication and its impact on financial markets.
- 2432 FRUCTUOSO BORRALLO, LUCÍA CUADRO-SÁEZ, CORINNA GHIRELLI and JAVIER J. PÉREZ: "El Niño" and "La Niña": Revisiting the impact on food commodity prices and euro area consumer prices.
- 2433 VÍCTOR CABALLERO, CORINNA GHIRELLI, ÁNGEL LUIS GÓMEZ and JAVIER J. PÉREZ: The public-private wage GAP in the euro area a decade after the sovereign debt crisis.
- 2434 LIDIA CRUCES, ISABEL MICÓ-MILLÁN and SUSANA PÁRRAGA: Female financial portfolio choices and marital property regimes.
- 2435 RODOLFO G. CAMPOS, ANA-SIMONA MANU, LUIS MOLINA and MARTA SUÁREZ-VARELA: China's financial spillovers to emerging markets.
- 2436 LUDOVIC PANON, LAURA LEBASTARD, MICHELE MANCINI, ALESSANDRO BORIN, PEONARE CAKA, GIANMARCO CARIOLA, DENNIS ESSERS, ELENA GENTILI, ANDREA LINARELLO, TULLIA PADELLINI, FRANCISCO REQUENA and JACOPO TIMINI: Inputs in Distress: Geoeconomic Fragmentation and Firms' Sourcing.
- 2437 DANIEL DEJUAN-BITRIA, WAYNE R. LANDSMAN, SERGIO MAYORDOMO and IRENE ROIBÁS: How do changes in financial reporting standards affect relationship lending?